

Exploring lived experiences and coping strategies employed by Black women in the SANDF post deployment-related trauma.

A mini dissertation submitted in partial fulfilment of the requirements for the degree.

MA Research Psychology

Department of Psychology, Faculty of Humanities
UNIVERSITY OF PRETORIA

Sihle Ntuli

Supervisor: Dr Sharon B. Sibanda

Submitted for Examination: December 2023



ACKNOWLEDGEMENTS

I would like to thank God for the courage, the strength and perseverance to go through this journey, it has not been an easy one.

To my supervisor Dr Sharon Sibanda, I'm very much grateful for your guidance, patience, and encouragement throughout this entire journey.

To my grandmother Thabile Dlungwane, I would like to say thank you for keeping me in your prayers always and encouraging me to reach for my dreams.

My dear siblings Andile Ntuli, Noxolo Zikalala and Xolisile Miya, thank you for your continued support, love, and care through and through.

To all my friends, there is a lot of you I cannot mention you one by one. I just want to say thank you for lifting me up when I was at my lowest and demotivated to continue through this journey. Not forgetting my in-charge nursing officer Capt. Mohlomi from CHA Centre, I will forever be grateful for your unwavering support.

To all my participants, thank you for availing yourself even in the midst of your busy schedules.

To my late family, my father Jabulani Ntuli, my mother Lindiwe Zikalala, aunt Khanyisile Ntuli and brother Sibusiso Ntuli, thank you for believing in me.



DECLARATION

I, Sihle Ntuli, hereby declare that this dissertation is my own work and that, where applicable, every effort has been made to correctly reference the work of other authors. Furthermore, I declare that this dissertation is to be submitted to the University of Pretoria and has not previously been submitted to this university or any other tertiary institution.

Signed on the 5th day of December 2023





ETHICS STATEMENT

I, Sihle Ntuli (u22754114), have obtained applicable Faculty research ethics approval for the research work titled 'Exploring lived experiences and coping strategies employed by Black women in the SANDF post deployment-related trauma on 15 November 2021(reference number: HUM010/0123). I, furthermore, declare that I have observed the ethical standards required in terms of the University of Pretoria's code of ethics for researchers and the policy guidelines for responsible research.



ABSTRACT

Soldiers encounter traumatic events during deployment that can have negative consequences on their mental and physical well-being. However, there is dearth of research studies that focuses on coping with trauma amongst the SANDF members following deployment, especially women. Therefore, the purpose of the current study was to explore the lived experiences of Black women in the SANDF and how they cope following deployment-related trauma. Data was collected by means of semi structured interviews among six participants who provided a rich detailed description of their lived experiences and coping strategies they used following a traumatic encounter. Interpretative phenomenological analysis was used to interpret and analyse the data gathered. The findings of the study illuminated participants' lived experiences of deployment, and the coping strategies they utilised following deployment-related trauma. The participants highlighted the emotional and psychological manifestations of their traumatic experiences, the role of their support system, as well as the barriers to seeking psychological services. They also highlighted positive aspects, shared lessons they learned from deployment and as well as challenges they were confronted with both during and post deployment. Given the complexity of trauma, it is therefore recommended that future studies look at the effectiveness of existing interventions put in place to combat the effects of deployment related-trauma in the SANDF, as well as the impact of deployment on the family of deployed soldier.



TABLE OF CONTENTS

ACKNOWLEDGEMENTS	ii
DECLARATION	iii
ETHICS STATEMENT	iv
ABSTRACT	v
CHAPTER 1	1
1. 1 Introduction	1
1. 2 Background	2
1. 3 Justification, Aim, Objectives, and Research Question	2
1. 3. 1 Justification	2
1. 3. 2 Aim	3
1. 3. 3 Objectives	4
1. 3. 4 Research question	4
1. 3. 4. 1 Primary research question	4
1. 3. 4. 2 Secondary research question	4
CHAPTER 2: LITERATURE REVIEW	5
2. 1 Introduction	5
2. 2 Military deployment	5
2. 3 Deployment related trauma	7
2. 3. 1 Post-traumatic stress disorder (PTSD) in the Military	8
2. 4 Psychological Debrief	10
2. 4. 1 The effectiveness of psychological debrief	11
2. 5 Women in peacekeeping	12
2. 6 Coping strategies	14
2. 7 Theoretical Framework	16
2. 8 Conclusion	17
CHAPTER 3: RESEARCH METHODOLOGY	18



3. 1 Introduction	18
3. 2 Qualitative Research Approach	18
3. 2. 1 Research design	19
3. 2. 1. 1 Interpretative Phenomenological Analysis (IPA)	19
3. 2. 1. 2 Limitations of the IPA.	20
3. 3 Sampling of Participants	20
3. 3. 1 Research concepts and participants	21
3. 3. 2 Selection criteria	21
3. 4 Data collection	21
3. 4. 1 Transcription	22
3. 5 Method of data analysis	22
3. 5. 1 Steps in Interpretive Phenomenology Analysis (IPA)	22
3. 6 Measures to enhance the quality of the research.	24
3. 6. 1 Trustworthiness	24
3. 6. 2. 1 Transferability.	25
3. 6. 2. 2 Credibility	25
3. 6. 2. 3 Dependability	25
3. 6. 2. 4 Confirmability	26
3. 8 Ethical procedures	26
3. 8. 1 Informed consent	26
3. 8. 2 Confidentiality and Anonymity	27
3. 8. 3 Emotional Risk	27
3. 7 Reflexivity	27
3. 7. 1 Personal reflexivity	28
3. 7. 2 Interpersonal Reflexivity	28
3. 7. 3 Epistemological Reflexivity.	28
3. 9 Conclusion	29



CHAPTER 4: FINDINGS	29
4. 1 Introduction	29
4. 2 Themes	29
4. 2. 1 Theme 1: PTSD symptoms	31
4. 2. 2 Theme 2: Coping Strategies	33
4. 2. 3 Theme 3: Support System	37
4. 2. 4 Theme 4: Lessons from deployment	38
4. 2. 5 Theme 5: Challenges	40
4. 2. 6 Theme 6: Barriers to seeking mental health services	44
4. 7 Conclusion	46
Chapter 5: Discussion	47
5. 1 Introduction	47
5. 2 PTSD symptoms	47
5. 3 Coping strategies	49
5. 4 Support systems	53
5. 5 Lessons from deployment	54
5. 6 Challenges	55
5. 7 Barriers to seeking mental health services.	57
5. 6 Recommendations	60
5. 7 Limitations	60
5. 8 Conclusion	60
References	62
Appendices	87
Appendix A- Written consent to participate in this study	87
Appendix B - Participant information sheet	87
Appendix C- Interview Schedule	87
Appendix D– Scripts of participants	87



Appendix E– Representation of master theme table	87
Appendix F– Extracts from scripts	87
Appendix G – Turnitin Report	87
Appendix H – Ethics Approval	88
List of tables	
Γable 1: Theme outline	29



CHAPTER 1

1. 1 Introduction

South Africa has the highest percentage of women participating in the United Nations (UN) Peace Keeping Operations (PKOs) in Africa (Moditsi & Gorur, 2020). A majority of contemporary UN peacekeeping missions operate in volatile and harsh conditions that are not easy to navigate (Di Raza, 2020). The implementation of the United Nations Security Resolution Council (UNSRC), meant an expansion of the peacekeeping mandate, thus making the operational environment increasingly challenging (George & Shepherd, 2016). The UNSRC focuses on the "operational and structural measures for the prevention of armed conflict, strengthening the rule of law, and promoting sustained economic growth, poverty eradication, social development, sustainable development, national reconciliation, good governance, democracy, gender equality and respect for, and protection of, human rights" (George & Shepherd, 2016, p. 270), which is more complex and multidimensional compared to the traditional peacekeeping that focused on the cessation of fire between fighting parties.

Post-deployment mental health issues are a source of potential concern for military health professionals around the world (Coetzee et al., 2010). This is due to physical and emotional stressors experienced by military personnel returning from deployment (Brenner et al., 2015). Following military deployment, returning home is frequently followed by significant changes in the environment for service members, including changes in the physical surroundings (such as the climate), mental state (such as feeling threatened), and the social milieu (Adler et al., 2011). As a result, post-deployment period introduces new stressors for service members and their families, especially those who are recovering from the effects of stressful and traumatic deployments (Marini et al., 2017). Although the effects of post-deployment stress on peacekeepers have been well documented, not much is known about the experiences of women, particularly Black women in peacekeeping after deployment-related trauma. Considering that the military setting is male-dominated, studies conducted tend to be more male-orientated and less focused on women's experiences of exposure to deployment-related trauma. Deployment-related trauma refers to traumatic events encountered during deployment, these include experiences such as combat exposure, exposure to potential danger or threatening situations, witnessing seriously ill or injured people, and witnessing people being killed (Frank et al., 2018).



1. 2 Background

In recent years, peacekeeping missions have evolved, and women have increasingly become part of peacekeeping operations. It is imperative that we acknowledge and understand the psychological well-being of women in service, in light of the increase in the percentage of women entering the force and participating in peacekeeping missions (Bertolazzi, 2010). The South African National Defence Force (SANDF) is among a very few military forces globally that incorporates women in a large scale in PKO's (Alchin et al., 2018). With the constant rapid expansion of roles and numbers of the SANDF women in these operations, a concern arises regarding the mental health of servicewomen post-deployment (Watson, 2019). Whilst there has been an upsurge of mental disorders such as posttraumatic stress disorder (PTSD), anxiety, and depression among peacekeepers, very few studies have focused specifically on women peacekeepers or have looked at the mental health outcomes among women secondary to long-term post-deployment adjustment (Di Razza, 2020).

Broader literature (Goldstein et al., 2017; Vogt et al., 2011) posits that women are more vulnerable to the effect of trauma exposure and present with a higher prevalence of PTSD. Furthermore, most studies (Morgan et al., 2017; Ochu et al., 2018; Shelef et al., 2018) available on coping strategies on PKOs have relied on limited samples regarding the effect of exposure to trauma on servicewomen, with analyses that do not directly focus on women's mental health post-deployment. SANDF servicewomen's increased exposure to trauma related to deployment missions offers an opportunity to evaluate mental health adjustment post-deployment. Therefore, the current study aims to explore the lived experiences and coping strategies of Black SANDF servicewomen post-exposure to deployment-related trauma.

1. 3 Justification, Aim, Objectives, and Research Question

1. 3. 1 Justification

The military is an organisation with a majority of tasks that require a certain level of physical strength which is often feasible for the male physique (Steidl & Brookshire, 2019). In addition, the military is a deploying organisation with members who can withstand the austere environment in both combat and peacekeeping operations (Adler et al., 2017). Women have proved to be suited to withstand such circumstances (Greer, 2017). However, a plethora of challenges arise with the integration of women in the military context especially



in the peace and security industry which have made it difficult for women to access opportunities for the longest time (Elnitsky et al., 2017).

Currently the military is more integrated than in the past. Women are part of leadership (Klenke, 2017), with a 28% representation of women within the SANDF (Martin, 2022). Furthermore, as of 2019 there were over 46 female brigadier generals compared to 2010 when the organisation had none (Helfrich, 2019). During the year 2019 the SANDF deployed 102 women in the Force Intervention Brigade (FIB) an operation that had no women previously (Helfrich, 2019). The organisation has not reached its target of 30%, however, progress made over the past decade is significant and not only in general positions but in senior positions as well as in peacekeeping operations (Martin, 2022). Women in the military still encounter challenges dealing with who they are and how they conduct or carry out their duties (Brown et al., 2021). Such attitudes and beliefs pose a threat to the integrity of the organisation and their mission or objectives (Trobaug, 2018).

It is therefore imperative that we look at the experiences of women within the force as they embark on new roles within an organisation entrenched within politico-historical masculine dominance. In addition, there have been studies of women within the military however those studies focused more on militaries in the Western countries (Belton, 2016; Gruber et al., 2021; Johnson et al., 2021; Markowitz et al., 2015; Warner, 2018; Welsh et al., 2015). A few South African researchers have given attention to South African peacekeepers (Harmse et al., 2022; Heinecken & Ferreira, 2012; Heinecken, 2020) however, none have investigated South African peacekeepers following deployment related trauma. There is little research on how women in the military adjust post-deployment; this could be because they make up a lower percentage of the force (Bertollozi, 2010). The dearth of studies conducted within the SANDF highlight essential gaps in psychological coping-related studies or research, especially with a focus on servicewomen. The current study, thus, aims to explore the lived experiences of Black SANDF women following deployment-related trauma, particularly their coping strategies.

1. 3. 2 Aim

The principal aim of this study is to explore the lived experiences of Black SANDF women following deployment-related trauma, particularly their coping strategies.



1. 3. 3 Objectives

The proposed study is guided by the following objectives:

- 1) To understand Black SANDF women's coping strategies in managing postdeployment-related trauma.
- 2) To explore the lived experiences of Black SANDF women post exposure to deployment-related trauma.

1. 3. 4 Research question

1. 3. 4. 1 Primary research question

What are the lived experiences of deployment-related trauma by Black SANDF women?

1. 3. 4. 2 Secondary research question

The secondary research question is as follows:

What positive and negative coping strategies do Black SANDF women adopt to cope with deployment-related trauma?

1. 5 Outline of chapters

Chapter 1 presents the introduction, background, justification, aims as well as the objectives of the study. This chapter also focuses on the research questions the study seeks to answer.

Chapter 2 provides a review of literature related to the experiences and coping strategies of Black SANDF service women and the theoretical framework that guides this study.

Chapter 3 discussed the methodology adopted in this research study. The discussion on the methodology provides information on the research design, sampling method, data collection and analysis of data. Furthermore, trustworthiness and ethical considerations of the study were discussed in this chapter.

Chapter 4 presents the findings of this study. This chapter provides a description and discussion the findings that emerged from the data gathered during the interviews. Therefore,



the themes and sub-themes that emerged from the data collected are discussed integrating the theoretical framework and reviewed literature. This aimed to describe the subjective experiences of Black SANDF women and how they cope with trauma.

Chapter 5 concludes this study with a summary of the findings. Followed by a discussion of the limitations the current and recommendations for future studies on deployment related trauma.

1. 6 Conclusion

This chapter presented an overview of the current study, as well as the background, justifications, aims, objectives and research questions that guide this study. Additionally, a brief outline of each chapter was presented. The following chapter is a review of literature related to lived experiences and coping strategies of Black SANDF women post deployment-related trauma.

CHAPTER 2: LITERATURE REVIEW

2. 1 Introduction

Military members encounter traumatic events that can have negative consequences on their physical and emotional wellbeing (Brooks, 2017). Women in service are a minor cohort and experience trauma due to the nature of the work their military responsibilities demand of them (Karim, 2017). In addition, they are confronted with a substantial amount of physical and psychological challenges whilst deployed (Bhakta & Logan, 2014). Exposure to life-threatening conditions during missions may lead to psychological adjustment difficulties post-deployment (Koopman &Van Dyk, 2012). The literature review will focus on the conceptualisation of military deployment, deployment-related trauma, post-traumatic stress disorder (PTSD) in the military, debriefing, women in peacekeeping and coping strategies within the military context. In addition, the theory of stress and coping is presented as the theoretical framework grounding this study.

2. 2 Military deployment

The SANDF has been engaged in the United Nations (UN) Peacekeeping Operations (PKOs) since 2001 (Bruwer & Van Dyk, 2005). The deployment of the SANDF in PKOs is part of South Africa's fulfilment of international obligations and a contribution to the stability and peace in the continent and other parts of the world (Mashatola & Bester, 2020). Military



deployments refer to the temporary transfer of military members from their home units to another location or country in support of combat or other military operations. (Biggs et al., 2017). Deployment of military members to foreign countries means separating from family, increased level of physical activity, changes in work habits, and responsibilities for a period of over a year (Rickovic, 2014). The concept of peacekeeping can be defined as the deployment of civilians, military, and police personnel by the UN in an effort to curb actual or potential conflict within or among states (Di Salvatore & Ruggeri, 2017). Peacekeeping operations are aimed at restoring peace and stability in areas experiencing conflict, this includes observing activities of all stakeholders involved, providing humanitarian and medical assistance as well as protecting the local population (Dervishi et al., 2015).

PKOs differ in terms of length, scale, outcome, and degree of difficulty (Bleckner, 2013). Peacekeepers encounter challenges such as exposure to environmental stressors which include poor living conditions, frequent exposure to poverty, illness, and physical injury that may contribute to their stress (Howard, 2019). The SANDF partakes in peacekeeping missions and peace enforcement missions, in which both men and women are deployed (Mashatola, 2022). According to Chapter 6 of the UN Charter "PKO's involve both military and civilian personnel who are delegated with the task of monitoring and assisting with the implementation of agreements reached between the belligerent parties, such activities are also mandated under the UN Charter; they take place with the consent of the conflicting parties and do not involve the use of force (other than in self-defence) by the peacekeepers" (Defense Review, 1998, p. 39). Chapter 7 of the UN Charter describes Peace-enforcement as "activities in the field, that require the use armed force in order restore international peace and security in situations where peace is threatened, where a breach of the peace occurs, or where there is an act of aggression (Defense Review, 1998, p. 39).

Peacekeepers may experience psychological distress manifesting in anxiety and depression which may result in alterations in experiential reality of self and the world (Gibbons et al., 2014). The programmes implemented in the SANDF focus on the predeployment phase, deployment phase, and post-deployment in an attempt to alleviate PTSD among soldiers during and post-deployment (Dhladhla & Van Dyk, 2009). The predeployment programme involves preparation training which includes providing psychoeducation on PTSD, building resilience, hardiness, morale, cohesion, and teaching on the importance of keeping communication with family members whilst deployed (Van't Wout & Van Dyk, 2016).



During deployment, the SANDF also deploys a multi-disciplinary team, consisting of a chaplain, social worker, psychologist, medical doctor, and nurses to assist soldiers to cope with stressors during deployment missions (Shinga, 2015). The deployment of a multi-professional team with a battalion may have a positive effect, as a study by Heinecken and Ferreira (2012) found that South African peacekeepers presented with low stress levels during deployment. The psychological debriefing programme implemented in the SANDF takes place immediately when soldiers return from peacekeeping operations, soldiers are screened for PTSD and are encouraged to share their emotional experiences of deployment (Dhladhla & Van Dyk, 2009). This is an effective way to assist or to ensure soldiers process their emotions following trauma exposure. However, it might not be feasible to detect PTSD immediately following a traumatic encounter as a significant body of literature points out that soldiers tend to experience delayed PTSD onset (Solomon, 2020).

Furthermore, when the operations end military members are not given enough time to transition back to their home life routine, which may be problematic or interfere with the adjustment process after deployment (Heinecken & Wilén, 2021). Factors associated with potential difficulties in adjusting to normal life routine include changes in emotions that stem from exposure to trauma, re-establishing relationships, financial concerns, and parenting (Vogt et al., 2011). A rapid transition may further contribute by not offering military members adequate time and opportunity to develop effective closure and perspective after the traumatic encounter (Nkewu & Van Dyk, 2016).

2. 3 Deployment related trauma

Soldiers who experience traumatic events during deployment are more prone to later report mental health problems, particularly PTSD or alcohol misuse (Waller et al., 2012). Trauma refers to an emotional response to a distressing event or series of events such as rape, accidents, or natural disasters (APA, 2022). Adverse reactions such as shock, and denial, are not uncommon after a traumatic event; long-term effects of trauma include unpredictable emotions, flashbacks, difficulties with relationships, and sometimes physical symptoms such as headaches or nausea (APA, 2022).

Deployment stressors can be divided into three categories, namely basic, cumulative, and traumatic stressors (Singh, 2020). Basic stressors refer to the daily deployment stressors such as boredom and ambiguity; these stressors are minor, and often associated with everyday routine that may cause irritability, anger, annoyance, as well as tension (Van't Wout



& Van Dyk, 2016). Generally, peacekeepers can overcome this type of stressor depending on their physical and psychological traits (Loscalzo et al., 2018). However, stress accumulates if not addressed, it can have negative effect on behavior and productivity of military members (Wisén et al., 2021). During peacekeeping missions, peacekeepers are normally given simple, monotonous, and repetitive work tasks; this may lead to boredom as there is a lack of entertainment (McCullough, 2016). Boredom becomes problematic when there is lack of meaning in professional work, as peacekeepers may believe that their job skills are deteriorating due to inactivity (Danieli, 2018). Ambiguity is another basic stressor, PKOs are inherently ambiguous (Rhoads, 2016). Peacekeepers reported on the ambiguity of the mission, this involves uncertainty in executing tasks as well as departure times (Bode & Karlsrud, 2019; Harmse, 2022).

Cumulative stressors refer to stressors that accumulate gradually, some problems might be significant and have prolonged effects while others might be minor, and a mere reflection of the challenges encountered daily; examples of cumulative stressors are isolation and danger (Hutton, 2020). Peacekeepers are usually isolated as they are deployed in remote areas where they struggle to keep communication with families and friends (Warner, 2018). Lengthy periods of isolation have an adverse impact on mental health (Gruber et at., 2021).

PKOs are characterised by a plethora of elements that pose dangers and threats to human life, these involve attack, injury, and contracting diseases (Howard, 2019). Exposures to the above conditions can affect adjustment and cognitive functioning (Howard, 2019). Critical traumatic stressor refers to "traumatic experience in which peacekeepers are exposed to physical or psychological assault in which there is a threat or harm to himself or herself or to the other peacekeepers" (Di Razza, 2020, p. 2). Examples of critical traumatic stressors include witnessing injuries, human remains or death of a coworker or the local population that peacekeepers are delegated to protect (Di Razza, 2020). Military members exposed to critical traumatic stressors are predisposed to resorting to alcohol as a coping mechanism and developing mental disorders (Wild et al., 2020).

Post-deployment mental health problems are influenced by both military and non-military factors, these include exposure to deployment-related trauma and previous mental health problems, in addition, they are related to exposure to potentially stressful deployment events (Moore et al., 2017).

2. 3. 1 Post-traumatic stress disorder (PTSD) in the Military



PTSD in the military is associated with potentially stressful experiences that pose a threat to the well-being of soldiers (Waller et al., 2012), in addition exposure to life threatening deployment events has been found to be a predictor of PTSD in deployed soldiers (Waller et al., 2012). PTSD is a psychological condition resulting from witnessing or experiencing single or multiple traumatic events; it is characterised by intense emotional and physical distress accompanied by severe anxiety, nightmares, flashbacks as well as avoidance of situations or places that trigger memories of trauma (APA, 2022; Han et al., 2014; Wastell, 2020). PTSD was also known as "shell shock" and combat fatigue during the era of World War I and II (Taylor-Desir & Association, 2022). PTSD symptoms usually commence within a period of three months following the occurrence of the traumatic event; however, symptoms may emerge at a later stage (Astill Wright et al., 2019).

During PKO's peacekeepers often encounter traumatic events that are akin to those of traditional combat operations (Visagie et al., 2022). Seedat et al. (2003) reported that 60% of South African peacekeepers experienced motor vehicle accident (MVA), 20% encountered life threatening injuries, and 22% experienced trauma of sudden death of someone close. Furthermore 26% of the peacekeepers met the DSM-V diagnostics criteria for PTSD; however very few of the peacekeepers sought psychological assistance (Seedat et al., 2003). The aforementioned study was supported by Sibanda (2020) in that SANDF members presented with undiagnosed PTSD symptoms due to trauma exposure resulting from previous and current military operations. Heinecken and Wilén (2021) investigated the challenges experienced by soldiers in adapting to the home environment after returning from PKO's among SANDF soldiers, they reported PTSD symptoms yet not many of the soldiers commented or elaborated on PTSD, although it was apparent in their discourse. The authors further contended that the inability or unwillingness of the soldiers to recognize PTSD symptoms may pose consequences for the military members and their families. Drawing from the above studies (Heinecken and Wilén, 2021; Seedat et al., 2003; Sibanda, 2020) it is evident that the SANDF requires a solid strategy to combat or reduce PTSD especially among peacekeepers post deployment.

According to the World Health Organization (WHO), there is a constellation of psychological disorders that occur following exposure to potentially traumatic events (PTE) (WHO, 2022). In addition, whilst there are over 50 mental health conditions linked with stress and trauma, PTSD is linked to many other mental health condition (Sareen, 2014).



Factors influencing how an individual appraises stress are dependent on the occurrence, frequency, and complication of the trauma (Alvers et al., 2020). Dervishi et al. (2015) posits that following experiencing traumatic encounter, it is challenging to predict who will develop PTSD symptoms, however there are certain risk factors that predispose peacekeepers to PTSD. Polusny (2013) identified three most common risk factors for PTSD among peacekeepers, namely unit support, coping strategies, and trauma exposure. Lack of social support plays a vital role in the development and maintenance of PTSD, particularly its emotional, cognitive, and behavioral elements (Simon et al., 2019). According to research lack of social support following trauma exposure leads to severe symptoms of PTSD (Anyaegbu et al., 2021; Moore et al., 2017; Wang et al., 2021). PTSD symptoms have been noted to be elevated post trauma exposure due to of lack of support among peacekeepers (Harmse et al., 2022; Heinecken, 2020).

Positive coping strategies have been found to buffer PTSD symptoms in peacekeepers, while negative coping strategies exacerbate PTSD symptoms (Kline et al., 2013; Possemato et al., 2014; Wright et al., 2013). Romero et al. (2020) noted that negative coping strategies were associated with PTSD symptoms among a cohort of U.S. military members post deployment. Negative coping strategies that have been noted mostly in military members include isolation, as well as the use of drugs and alcohol (Armenta et al., 2018; Matshayisa & Letsosa, 2019; Mbhele, 2021; Nkewu & Van Dyk, 2016).

As stated previously peacekeepers encounter a variety of atrocities which may result in the development of PTSD symptoms (Brenner et al., 2015). However, some countries attested that most of their peacekeepers report low symptoms of PTSD post deployment; this may be due to training offered prior to and post deployment (Álvares et al., 2020; Brenner et al., 2015; Doody et al., 2022; Warner, 2018).

2. 4 Psychological Debrief

The SANDF engages in combat, peacekeeping operations, and disaster management. Peacekeepers are responsible for protecting people living in conflict zones and are subjected to traumatic encounters (Dhladhla & Van Dyk, 2009). The nature of such missions involves witnessing violent fighting, killing of women and children, child soldiers and their own forces; these events may contribute to members developing PTSD (Halterman & Irvine, 2014). To mitigate the effects of trauma within the SANDF, a once-off debriefing session takes place with members, facilitated by psychologists immediately after the occurrence of an



incidence as well as after deployment at the demobilisation area (Dhladhla & Van Dyk, 2016) Psychological debriefing, which is employed by many military forces worldwide, is one of the most popular early interventions with military populations (Adler et al., 2009). Psychological Debrief can be defined as a technique that enables survivors to process and reflect on the traumatic events they have witnessed while also gaining personal control over the situation (Arancibia, 2022).

Three reasons led to the development of psychological debriefing in the military: to alleviate the psychological effects of traumatic events, to ease acute response, and to decrease the occurrence of PTSD (Snowdon, 2021). However, much controversy revolves around the effectiveness and neutrality of psychological debrief, while some authors argue that it reduces trauma symptoms (Greenberg, 2007); others contend that it can be detrimental in that it can impair psychological adjustment as a whole and have negative psychological effects (Wood et al., 2023; Wright et al., 2013). On the other hand, Van Dyk and Dhladhla (2009; 2016), support the importance of holistic management, and implementation of educational military programmes before and after deployment, including hosting a debriefing session immediately after deployment to alleviate the prevalence of PTSD in the SANDF.

2. 4. 1 The effectiveness of psychological debrief.

The effectiveness of psychological debriefing has been questioned for more than two decades. Several authors stated the positive and negative aspects on psychological debrief. Feuer (2021) asserted that psychological debrief can be used as a preventative measure, and proper usage of the intervention may also screen members who are suffering from acute stress. While other authors (Arancibia et al., 2022; Kearns et al., 2012; Rose et al., 2002; Vignaud et al., 2022) argued that it has limited efficacy and is potentially harmful.

2. 4. 1. 1 Positive aspect of psychological debrief. Adler et al. (2011) conducted a study to investigate whether soldiers who received psychological debrief would show minimal psychological distress, and present with better physical health compared to those who did not receive psychological debrief. Results demonstrated that soldiers who received psychological debrief were coping well, presented with low levels of post-traumatic stress and were physically healthier in comparison to the soldiers who did not receive psychological debrief. Castro et al. (2012) compared psychological debrief with three other early interventions in a sample of U.S. soldiers. Results revealed that all soldiers who received psychological debrief presented with fewer symptoms of PTSD and sleep problems in



comparison to those soldiers who received three other interventions. The authors concluded that psychological debrief might be a buffer against PTSD symptoms (Castro et al., 2012).

In a study conducted by Tuckey and Scott (2014) to investigate the effectiveness of psychological debrief among emergency care workers, results demonstrated that the healthcare workers who had received psychological debrief reported less use of alcohol; thus, indicating that psychological debrief was associated with less alcohol use and good quality of life compared to the emergency workers who did not receive psychological debrief. However, there was no significant effect on PTSD symptoms and emotional distress in this study; the authors concluded that debrief may be beneficial for broader functioning post exposure to traumatic events (Tuckey & Scott, 2014).

2. 4. 1. 2 Negative aspects on psychological debrief. The most common pitfall stated about psychological debrief, was that it exacerbates trauma symptoms of victims (Rose et al., 2002). Some authors (Gist & Woodall, 2013; Tuckey & Scott, 2014; Seelandt et al., 2021). contended that, it is unreasonable to expect a single or once off session intervention to prevent a complex psychopathological disorder such as PTSD. Others argued that psychological debrief may form an important part of a comprehensive post incident intervention, but it is not to be considered solely as an intervention (Hawker et al., 2011).

Vignaud et al. (2022) conducted a meta-analysis study to investigate the effectiveness of psychological debrief to prevent PTSD among a cohort of soldiers and emergency workers; very few studies reported on the psychological debrief's effectiveness to decrease PTSD symptoms such as flashbacks, avoidance, and isolation. Most of the articles in the systematic review did not support the effectiveness of the psychological debrief to reduce PTSD symptoms; the authors concluded that the early intervention may be potentially harmful to those who receive it (Vanguard et al., 2022). This study was supported by Rose et al. (2002) in a meta-analysis that demonstrated that debriefing can be potentially harmful.

2. 5 Women in peacekeeping

Over the past decade, the United Nation Security Council (UNSC) established a resolution to factor in gender equality in the PKOs in order to improve gender balances within the security sector (Vermeij, 2022). There has been significant progress towards achieving this resolution (Gianni & Vermeij, 2014). Although there's been significant progress in stabilising gender equality and gender mainstreaming in the PKOs, several challenges still exist in the implementation of the resolution. The UN policies are restricted



when it comes to state or stakeholders involved in the PKOs as some states involved appear to be passive in embracing the resolution (Gianni & Vermeij, 2014).

The UNSC resolution has been viewed as a benchmark as it legally binds states and stakeholders to include gender aspects in their peacekeeping processes (Barrow, 2016). The resolution is mostly understood as an establishment of the four fundamentals guiding the gender mainstreaming policy, namely prevention, protection, participation as well as recovery (Barrow, 2016). The first fundamental element demands interventions that decrease gender-based violence and sexual related conflict in PKOs (Karim & Beardsley, 2017). The second fundamental advocates for an increase in participation of women in decision making roles tasks in all levels and stages during conflict management (O'Reilly et al., 2015). The third fundamental guideline, concerning protection advocates for the support of women, children, as well as vulnerable and marginalised cohorts, especially those affected by sexual related conflicts and gender-based violence (Sharland, 2021).

Lastly, the fourth fundamental guideline related to relief and recovery, advocates for gender perspective to be incorporated in all phases of recovery from conflict, this includes long term availability or accessibility of healthcare and other necessary service (Luedke et al., 2017). The SANDF increased the percentage of women involved in peacekeeping to implement the UNSC resolution (Karim & Beardsley, 2017). Karim (2017) asserted that increasing women peacekeepers has the potential to create a positive impact on local women, inspiring them to engage in the political security affairs of their country. On the other hand, Wilén (2020) contends that gender mainstreaming or resolution may place high expectations and a greater burden on women peacekeepers and should be abandoned altogether.

Women are deployed in various capacities, including policing, military, or civilian obligations, and have a positive influence on the conditions of peacekeeping, especially by emphasizing women's roles in building peace and upholding women's rights (UN, 2012). Women in peacekeeping have shown that they can carry out the same responsibilities, uphold the same standards, and work under the same difficult situations as their male counterparts in all areas of peacekeeping (UN, 2012). It has been established that greater representation of women in peacekeeping operations increases mission effectiveness, ensures better access to local people, especially women, and more effectively promotes human rights and the protection of civilians (Basu, 2018; Helfrich, 2019). According to Karim & Beardsley (2017, p.62) "Women peacekeepers operate within and respond to this gendered environment,



providing a "humanising" dimension to peacekeeping, often undertaking activities that would be considered inappropriate for men".

The deployment of women peacekeepers at checkpoints during PKOs has been attributed to creating a less combative environment in extremely volatile security environments (Sharland, 2019). In addition, women have great interpersonal skills than their male counterparts and gain access to intelligence from different members of society with ease, thus obtaining a more holistic view of the security environment (Ghimire, 2017). The UN had its second woman as a force commander since its inception (Ghittoni, 2018). The SANDF had its first woman as a battalion commander to lead a battalion successfully in 2019 (Martín de la Rosa & Lázaro, 2019). In light of these progressive developments, women peacekeepers have proven that they are valuable, efficient, and contribute significantly to PKOs.

Despite the successful integration, women still face many challenges within the security environment. The three most common challenges encountered by women peacekeepers are exclusion, discrimination, as well as sexual harassment and assault (Bleckner, 2013; Kreutz & Cardenas, 2017; Newby & O'Malley, 2021). Women are usually excluded from decision-making matters and leadership positions during the PKOs (Owuor, 2021). In addition to facing exclusion, women tend to be stereotyped as more nurturing and less threatening, and lacking physical strength and endurance, especially during patrols (Alchin, 2015). Furthermore, women are regarded as posing a gender security risk, particularly in hypermasculine contexts such as PKOs, where they are seen as sexual objects thus facing the threat of being sexually harassed or assaulted, as well as a threat to existing gender power relations that affect male dominance (Sharland, 2021). It is thus pivotal to understand how servicewomen cope with the myriad of potentially traumatising challenges during and post deployment missions.

2. 6 Coping strategies

Traumatic life events shatter our assumptions about ourselves and our world, in the aftermath of these extreme experiences, coping is necessary which involves a demanding task of reconstructing our assumptive world; a task that entails an intricate balance between confronting and avoiding trauma-related feelings, thoughts, and images (Janoff-Bulman, 1989; Nygaard & Heir, 2012). Positive coping strategies form an integral part of mental health and play a critical role in regulating response to stresses (Zhao et al., 2020). Specific



coping strategies mitigate stress and endorse positive psychological outcomes, while others worsen stress and promote negative psychological outcomes (Smith et al., 2016). However, the effectiveness of any given coping strategy is also dependent on resiliency and social support (Smith et al., 2016). Folkman and Lazarus (2013) referred to coping as cognitive and behavioral responses of an individual in confronting and managing stressful situations, however not all forms of coping are considered beneficial. Coping strategies can be defined as mechanisms people use when they encounter stress and/or trauma to assist in managing painful or difficult emotions (Algorani & Gupta, 2022)

Coping strategies can assist people in adjusting to stressful events while maintaining their emotional well-being (Algorani & Gupta, 2022). Employing coping strategies, such as self-blame or avoidance to deal with stressors leads to negative results (Nzabamwita, 2017; Mbhele, 2021). Thus, coping strategies may be categorised into positive and negative coping. Positive coping is associated with rational problem-solving, while negative coping is associated with avoiding, withdrawing, or denying the problem (Martos Martínez et al., 2021).

Lazarus and Folkman (1984) identified two types of coping strategies, emotional-focused and problem focused coping strategies. In the absence of positive coping strategies peacekeepers are more likely to adopt negative coping strategies (Britt, 2017). Emotion-focused coping strategies refer to coping strategies that involve controlling emotional response to stress (Flood & Keegan, 2022). Alcohol misuse and avoidance are emotion-focused coping strategies that are common among peacekeepers post deployment (Britt et al., 2015; Castro et al., 2012; Soir, 2017; Platania, 2022). Avoidance coping strategies are associated with negative traits such as frustration, anxiety, and depression. In addition, they are perceived to be ineffective in solving problems (Smith et al., 2016; Stanisławski, 2019). Military members with low resilience tend to adopt avoidance coping strategies (Bartone, 2020). Alcohol misuse has been linked to mental health problems such as PTSD in peacekeepers (Dirkzwegar et al., 2005; Souza et al., 2011). The use of alcohol as a coping mechanism is more prevalent in peacekeepers who have witnessed critical trauma such as the death or witnessing human remains and women peacekeepers who have experienced military sexual trauma (Schumm & Chard, 2012).

Problem focus coping involves the use of coping strategies that are intended to impact or alter stress through direct action, problem solving and active decision making (Biggs et al.,



2017). Problem focused coping strategies common among peacekeepers are seeking social support and generating alternative solutions. (Di Salvatore & Ruggeri, 2017; Harney & Leeman, 2021; Yavnai et al., 2018). Seeking social support is a positive problem focused coping strategy which involves talking to someone regarding the problem (Stanisławski, 2019). A study conducted by Heinecken and Ferreira (2012) among South African peacekeepers in the Democratic Republic of Congo (DRC) evidenced that speaking to someone regarding your stressors can lower your stress levels. Generating alternative solutions involves evaluating the problem, which may assist peacekeepers cope with stressors and lower levels of anxiety and depression (Morgan et al., 2017). Lack of effective coping strategies can result in emotions such as hopelessness, powerlessness as well as suicidal ideation (Rajappa et al., 2012).

Research studies found that soldiers experience difficulties in readjusting back to normal life after deployment (Balderrama-Durbin et al., 2018; Elnitsky et al., 2017; Heinecken & Ferreira, 2012). According to Bryant et al. (2021, p.63) "this difficulty in readjustment may, in turn, lead to the onset of PTSD". Furthermore, positive coping strategies have the potential to decrease symptoms of PTSD and facilitate growth following deployment trauma exposure in peacekeepers (Wood et al., 2023). However, there is a dearth of research that demonstrates the type of coping strategies employed by women peacekeepers following deployment-related trauma exposure, a gap that the current study aimed to address.

2. 7 Theoretical Framework

The theoretical framework that underpins the findings of the current study, is the theory of stress and coping (Lazarus & Folkman, 1984). The transactional model of stress which is also known as the theory of appraisal, stress, and coping; provides clarification on how individuals evaluate and respond to stressful events. In their seminal work, Lazarus and Folkman (1984) coined the term "coping" to define stress responses. Lazarus and Folkman (1984, p.19) defined "stress as a condition or feeling that occurs when a person perceives an event as "exceeding his or her resources and endangering well-being." This theory proposed that the optimal way to assess coping was through observing how a person appraises the stress or problem. Personal appraisal refers to the various ways in which people attempt to change negative aspects of their lives to reduce the internal threat of stressors; an appraisal can be a primary or secondary perception of a stressor, and thereby an evaluation of potential effectiveness and consequences of coping behavior (Lazarus, 1993). The theory will illustrate



how Black women in the SANDF appraise and cope with deployment related stress during and after deployment.

According to this theory, coping is effective if stress is accurately appraised and specific behavioral and cognitive strategies are used to manage, reduce, or tolerate stressful events (Brown et al., 2021). Lazarus and Folkman (1984) identified two types of coping styles: emotion-centered coping, which seeks to regulate internal emotions and the meaning of an event, and problem-based coping, which seeks to change the problem or conflict; the authors further investigated stress and coping in different contexts. Researchers have utilised the theory to identify stressors as well as coping strategies, both negative and positive (Castro et al., 2017; Heff & Willoughby, 2017; Vahedi et al., 2021). The theory will further elucidate on how they attempt to reduce the stressors encountered with the resources at their disposal and how they measure the effectiveness of such resources as well as the coping strategies used and the consequences thereof. The theory will also shed light on the type of coping strategies employed by the servicewomen which can either be negative or positive.

Lazarus (2000) posits that when an individual experiences stress, they initially appraise whether the stress is a threat, or a potential source of harm or loss. After appraising the stressor, emotion focused, or problem focused coping strategies emerge (Folkman & Lazarus, 2013). The type of coping strategies adopted by an individual is dependent on how the stress is appraised - threat, challenge or source of potential harm or loss (Biggs et al., 2017). Emotion focused coping strategies are viewed as negative coping (Folkman, 1994). Negative coping strategies include avoidance and the use of alcohol (Algorani & Gupta, 2021). In a military context negative coping can manifest in military members' excessive consumption of alcohol following exposure to traumatic events in deployment and avoiding getting psychological assistance (Brenner et al., 2015). Problem focused coping strategies are deemed to be positive coping strategies and might help decrease stress (Lazarus, 2020). The theory aligns with the aim of the present study in that it shed light on the experiences of how Black SANDF women cope, thus illuminating the types coping strategies they employ post-exposure to deployment-related trauma.

2. 8 Conclusion

Chapter 2 presented a review of literature related to the lived experiences of service women in the SANDF following deployment-related trauma which focused on military deployment, debriefing and coping strategies. The impact of deployment-related trauma on



peacekeepers was discussed as well as the role of women in peacekeeping. PTSD risk factors in the military context was elaborated on, including coping strategies employed by peacekeepers. The theoretical framework grounding this study, the theory of stress and coping was also discussed.

CHAPTER 3: RESEARCH METHODOLOGY

3. 1 Introduction

This chapter discusses the research methodology that guides this study. The qualitative approach was deemed suitable for this study. Data was collected by means of semi- structured interviews. Interpretative phenomenological analysis (IPA) was adopted to analyse data collected from the interviews. This chapter also discusses measures that ensured trustworthiness of this study, as well as ethical considerations and concludes with a section on reflexivity.

3. 2 Qualitative Research Approach

A qualitative research approach was deemed suitable by the researcher in order to gain insight into the lived experiences and coping strategies of Black women in the SANDF post deployment-related trauma. Qualitative research entails gathering and analysing non-numerical data to better comprehend experiences, viewpoints, or ideas (Bells & Waters, 2018). It can be utilised to uncover intricate details about a phenomenon or discover new research concept (Alsaigh & Coyne, 2021). Qualitative research dates back to the early part of the 19th century, well before quantitative statistical techniques were developed, and is rooted in anthropology, sociology, psychology, linguistics, and semiotics (Banks, 2018). The goal of qualitative research is to increase or widen our understanding of how things in our social world come to be the way they are (Hancock et al., 2021). Furthermore, an important focus of the qualitative approach is the emic perspective, that is the perspective, meaning, and interpretation of the individual involved in the research (Erciyes, 2020).

Qualitative research approaches rest on the assumption that reality is socially constructed by humans which can be altered and understood subjectively (Ngozwna, 2018), which is contrary to the objective quantitative approach (Rahman, 2016; Willig, 2013). Denzin and Lincoln (2011) asserted that qualitative research seeks to describe participants' perspective or point of view. As a result, qualitative oriented approach aims to facilitate the



comprehension of those perspective, in a metaphorical sense, these perspective serves as a compass of qualitative research, as they provide a solid foundation for researchers to gain insight into how human experience the phenomena under investigation (Alase, 2017).

Therefore, the perspective of lived experiences provides a strong foundation for the researcher to establish conclusions regarding how the phenomenon was experienced. This was essential for the current study since the researcher sought to gain an understanding of how Black SANDF women experienced deployment-related trauma, particularly the coping strategies they employed.

3. 2. 1 Research design

A phenomenological approach premised on qualitative research will be the guide principle of the present study. In this section, the researcher explains the main assumptions of the interpretative phenomenology analysis approach.

3. 2. 1. 1 Interpretative Phenomenological Analysis (IPA). IPA comprises three theoretical perspectives namely phenomenology, hermeneutics and idiographic. Phenomenology seeks to explore the essence of lived experiences, which is crucial to gain in depth understanding of the phenomenon under investigation (Ramsook, 2018), In the current study, the lived experiences and coping strategies of deployment-related trauma among Black SANDF women was explored. Hermeneutics focuses on exploring lived experiences of the participants putting emphasis on individual interpretation of participants within a specific context (Hopkins et al., 2017). This theoretical perspective provides a rich description of the participant's experiences which are usually taken for granted when researchers attempt to make meaning and gain insight of their lived experiences (Alsaigh & Coyne, 2021). Researchers attempt to understand how participants make sense of their experiences; this is known as double hermeneutic (Pietkiewicz & Smith, 2014). Through the hermeneutic perspective, the researcher was able to describe the experiences of Black service women and how they coped following deployment-related trauma, as well as how they navigated other deployment challenges.

The ideographic perspective is concerned with studying a single case intensively to gain an in-depth understanding about a particular case or participant (Miller et al., 2018). Participants experiences were looked at individually to gain more insight regarding their experiences, this was possible as the sample was small. The ideographic perspective seeks to describe a particular individual and places emphasis on the individual's traits or unique trait.



This involves examining the uniqueness of the individual's behavior and adjustment aspects rather than producing a universal or overall set of psychological constructs that may apply to a particular population (APA, 2022).

3. 2. 1. 2 Limitations of the IPA. While IPA is considered to provide practical, accessible, and easy to understand guidelines in terms of understanding and interpreting participants experiences, it is not without criticism (Love et al., 2020). The approach has been criticised for its lack of standardisation as well as its ambiguity (Turffor, 2017). Other scholars contended that it is rather more descriptive and inadequately interpretive (Motta & Larkin, 2023). Another concern raised was that of the accuracy of IPA in capturing lived experiences and meaning participants make thereof, whether it captured meaning or it's opinions thereof (Turffor, 2017).

Phenomenology seeks to understand nuanced accounts of individuals experiences; however, the approach relies on the intersubjectivity between participants and experience of the researcher (Neubauer et al., 2019). This necessitates good communication skills from both the participants and the researcher to successfully capture experiences in-depth. Therefore, the approach may be seen as reserved and accessible to individuals who are eloquently and fluently enough to describe their experiences (Murray, 2023).

3. 3 Sampling of Participants

The participants were selected through a purposive sampling method from various operational units. Prior to the commencement of the study, permission was requested from the Defence Intelligence (DI) through a formal letter (Appendix C). Permission was also be obtained from the Officers Commanding of the three identified operational units in Gauteng to access participants (Appendix D), the sample size is n= 6. Once permission was obtained, the study was advertised through word of mouth, meaning the researcher approached identified operational units, talked to potential participants, and directly explain the nature and purpose of the study. Babbie (2013) defines purposive sampling as an approach in which the population to be observed or investigated is selected based on the researcher's judgement about which members will be most suitable or representative. The sampling technique was deemed suitable, because the study comprised a particular population that is not commonly found. For the current study, the population was Black women who are uniformed members in the SANDF. According to the statistical data on the race and gender breakdown of the SANDF as of 2021, there was 68,9% males and 31,1% females, of which 75,8% were Black



African, 10,8% white, coloured 12,2% and 1,1% Indian (Janse Van Rensburg, 2021). As stated in the introduction most military studies conducted within the military are maleoriented and thus the focus in this study is on women, specifically ethnically Black since they constitute a higher percentage in the SANDF and thus partake in PKO's. The SANDF consists of four arms of service, namely, the South African Army (SA Army), South African Air Force (SAAF), South African Navy (SAN), and South African Military Health Service (SAMHS). For the present study, the sample comprised six Black women officers and non-commissioned officers from the SA Army (Infanteers) and SAMHS (medical personnel) who have previously deployed to Operation Mistral and Operation Vikela.

3. 3. 1 Research concepts and participants

Recruitment of participants took place in the province of Gauteng. One of the smallest among the nine provinces of South Africa situated in the Northeastern region of the country. The province contains three metropolitan municipalities, Johannesburg, Tshwane metropolitan and Ekurhuleni municipality; as well as two districts municipalities that are divided into six local municipalities. The sample used in the study comprised women from operational units in Gauteng.

3. 3. 2 Selection criteria

A selection criterion was developed to ensure that eligible candidates are selected for the study. Establishing a selection criterion is important in that it provides pivotal attributes of the target population that the researcher will use to answer the research question (Nowell et al., 2017). The primary research question that this study sought to answer was what Black SANDF women's lived experiences was of possible trauma related to deployment, therefore participants were required to meet the following criteria to be included: Be a Black service woman in the SANDF, deployed externally (Mozambique or DRC), experienced deployment-related trauma, returned from deployment in the past 24 months, be an operational medic or infantry personnel. Participants were excluded based on the following criteria: male, returned from deployment more than 24 months ago, deployed internally, non-black service women, deployed for less than six months.

3. 4 Data collection

Semi-structured in-depth interviews were used in the current study to collect data with six participants. The utilisation of semi-structured interviews allowed the researcher to



collect open-ended data, explore participants thoughts, and feelings of their lived experiences and how they cope after deployment-related trauma (Dejonckheere & Vaughn, 2019). All participants signed consent forms. Interviews were approximately 30 to 45 minutes in length. The purpose of the research, confidentiality, as well as the right to withdraw was reiterated. The semi-structured interview guide allowed room for both flexibility and probing, to gather the necessary information from participants (Qu & Dumay, 2011). Questions from the semi-structured interviews guided the in-depth information gathered. Examples of these open-ended questions included: "How would you describe some of the deployment-related trauma you have experienced?" "What are some of the coping strategies that have been beneficial and not so beneficial for you?" The interviews were audio recorded, with consent from the participants to assist the researcher with the transcription process. The participants were thanked and given both the researcher's and supervisor's contact details should they have any questions following the interview.

3. 4. 1 Transcription

Phenomenological researchers tend to utilise linguistic data (Smith and Osborne, 2006). After the interviews were completed, each interview recording was transcribed verbatim. Each case was listened to repeatedly to fully capture the meaning the participants ascribed to their experiences. Verbal and nonverbal cues that were observed during the interviews were also captured.

3. 5 Method of data analysis

3. 5. 1 Steps in Interpretive Phenomenology Analysis (IPA)

The proposed study used Interpretive Phenomenology Analysis (IPA) to capture the lived experiences and coping strategies employed by Black SANDF women post deployment-related trauma, which is in line with the interpretive phenomenological approach that underpins this study. The benefit of using this methodology is that it offers the researcher an opportunity to grasp the innermost deliberation of the participants lived experiences. In addition, as a participant-oriented approach IPA enables the participants to narrate their lived experiences in a manner they deem fit (Alase, 2017). The current study was guided by the six steps of IPA as defined by Smith and Osborne (2009).

Step 1. The first step of the analysis involves familiarising oneself with the data, listening to the recording, reading, and re-reading the transcripts multiple times (Smith &



Osborne, 2015). This was aimed at helping the researcher to immense herself in the data as well as to ensure that each case is studied in-depth. (Mohajan, 2018). During the repeated readings of each transcript, the researcher actively engaged with the data by making notes of her initial impressions in a journal (Smith et al., 2009). At this stage of the analysis, the researcher's notes were unfocused and wide-ranging (Willig, 2013). As a part of the ideographic element of IPA, she made sure to focus her attention on each individual transcript before moving onto the next transcript and thereafter, to the cross-case analysis (Willig, 2013).

Step 2. Initial noting. During this step the researcher attempts to make sense of the language the participants make use of when describing their lived experiences. This step required the researcher to make notes on the script based on what she observed and reflect on the interviews. The researcher identified any potential bias, reflected on perceptions and views regarding the topic (Smith et al., 2016). After several re-readings of each transcript, the researcher made notes on the transcript using the comments tool on Microsoft Word. Initially, she made more descriptive comments by trying to stay as close to the ways in which participants described their experiences.

Later, she added notes on a more interpretive level. Specifically, this involved making sense of the language that participants used to describe their experiences. Additionally, on a more conceptual level, she explored the reasons underlying participants' experiences (Smith et al., 2009; Willig, 2013). She recorded some of her feelings in response to the transcripts in her journal. This information was useful to identify her own biases and to provide insight into the participants' experiences. In her journal, she also recorded similarities, differences, and contradictions in what the participants were saying.

Step 3. The third step of IPA involved reading notes on the transcripts in order to get emergent themes. In this phase of analysis, the researcher read her notes on the transcript with the aim of grouping together various ideas to establish themes. She also started to add potential labels for each theme in the comments section.

Step 4. Look for connections across emergent themes. In this step the researcher looked for patterns between the clustered themes and then listed the emerging themes and looked for connections (Smith, 2011). Thereafter, she typed out a list of the themes in a Microsoft Word document. She started to group some themes together to form clusters of related themes. This helped to start delineating between main themes and sub-themes (Smith



et al., 2009). The researcher used the guideline provided by Smith et al. (2009) to assist with grouping together main themes into sub-themes.

Step 5. Moving on to the next case. This step involves bracketing previous themes and keeping an open mind to do justice to the individuality of each new case. Repeated steps 1-4 for all the participant and adhered to the idiographic perspective of the IPA (Gill, 2020). The researcher completed steps 1-4 for all six transcripts.

Step 6. Searching for patterns across cases. The final step of IPA is concerned with writing up narrative accounts of the participants utilising themes elicited. Finding patterns, connections, and differences within the data set. This included combining themes, deleting some and creating new themes (Smith et al., 2009). At this stage the researcher had developed a document highlighting the main and sub-themes per interview. Thereafter, she searched for connections, patterns as well as differences across the interviews (Card, 2017). During this stage of analysis, she used the techniques presented by Smith et al. (2009) to combine some themes, delete other themes and create new themes (Smith et al., 2009). She began to develop a table in which she highlighted potential main themes and sub-themes across interviews. It took several drafts to finalise the table of themes. The researcher concluded the analysis by rereading all transcripts to ensure that all the participants' experiences were included in the findings.

3. 6 Measures to enhance the quality of the research.

In order to evaluate the study's quality, it is imperative that the researcher adheres to the four tenets trustworthiness namely, transferability, credibility, dependability, and confirmability (Stenfors et al., 2020).

3. 6. 1 Trustworthiness

"Trustworthiness" in qualitative research refers to the quality, authenticity, and truthfulness of the findings. It is a measure of the degree of trust or confidence that readers can have in the research results (Cypress, 2017). Trustworthiness in qualitative research refers to the methods used to ensure that the research process has been carried out correctly, and that the findings are reliable, valid, and credible (Hancock et al., 2021). Trustworthiness is both a goal of the study and a criterion for judging the quality of the research. From the phenomenon of trustworthiness, they derived four terms within the naturalistic paradigm to replace the rationalistic terms specifically, credibility, transferability, dependability, and confirmability (Cypress, 2017).



3. 6. 2. 1 Transferability.

Transferability of research finding refers to the extent which it can be applied in other context and studies (Johnson et al., 2020). Contrary to generalisability which applies to specific types of quantitative approaches, transferability is applicable to several types of research approaches. Through thick descriptions of the research, the researcher paints a picture of how certain life events unravel in a particular context, providing a better understanding of the phenomenon under investigation (Tracy & Hinrichs, 2017). Thus, allowing the reader to judge whether the study can be transferred or applied in a different context.

This study ensured transferability by providing thick descriptions of the participants lived experiences, as well as the research method the researcher utilised during data collection and analysis. Providing a rich description of participants' responses (and the researcher's interpretations) will ensure transferability, it will also ensure that the experiences of Black women in the SANDF is understood in-depth and how it can be applied in other settings.

3. 6. 2. 2 Credibility

Credibility is one of the key components of trustworthiness in qualitative research, which can be defined as the confidence that can be placed in the truth of the research findings (Stahl, 2020). Credibility is established by demonstrating that the research findings represent plausible information that has been drawn from the participants' original data, and that the interpretation of the data accurately reflects the participants' original views (Lincoln & Guba, 1985). This study used member checking to ensure credibility, this was done by confirming with the participants whether their experiences on the interview scripts were captured and analysed correctly following data analysis. McKim (2023) defined member checking as the process of confirming interview transcripts or analysed data with participants whether their experiences were captured accurately. One way in which the researcher increased credibility was by reading extensively about trauma within the military context to ensure that she had sufficient understanding to accurately represent participants' perceptions of the deployment-related experiences of traumatisation (Atkinson, 2008).

3. 6. 2. 3 Dependability



To attain dependability, researchers can ensure the research process is logical, traceable, and well documented. Dependability is another key component of trustworthiness in qualitative research, and it refers to the consistency and stability of the research findings over time and across different researchers and contexts (Hancock & Algozzine, 2017). To achieve dependability, the researcher kept an audit trail by describing in detail about the decisions taken in the research process from the commencement of the project right up until the development of themes and discussion of the findings (Korstjen & Moser, 2018).

3. 6. 2. 4 Confirmability

A fourth perspective on trustworthiness is confirmability or getting as close to objective reality as qualitative research can get (Stalh & King, 2020). Confirmability refers to the degree to which the research findings represent the experiences and ideas of the participants, rather than being influenced by the characteristics or biases of the researcher (Algozzin & Hancock, 2017). It is often seen as the qualitative equivalent of objectivity, which is a key concern in quantitative research (Alase, 2017). As afore mentioned, the study is grounded in phenomenology which assumes that the researcher shapes the research project through their interpretation of the findings (Bauman, 2010; Dahlberg & Dahlberg, 2004). In order to ensure confirmability, the interpretation must be grounded in the data and the nature of the interpretation must be intersubjective (Korstjens & Moser, 2018). To ensure confirmability, the researcher took steps to minimise her own influence on the research process and findings. Dependability of this study was ensured by discussing the interpretations with the supervisor and making use of peer reviews as well as member checking. In addition, an external researcher was requested to analyse the data independently in order to confirm whether the findings were consistent and can be repeated.

3. 8 Ethical procedures

3. 8. 1 Informed consent

According to Denzin and Lincoln (2011), informed consent is the corner stone of ethical research. This ethical consideration comprises two components, each component requires careful consideration, specifically informed and consent. Participants were fully informed about what is required of them and how the data collected will be utilised as well as the risks involved in the study. Badampudi et al. (2022), asserts that participants must provide explicit active signed consent to participating in the study, which includes a clear



understanding of their rights to access the information they gave and the right to withdraw at any given point in time when they wish to do so.

3. 8. 2 Confidentiality and Anonymity

The concept of anonymity requires the researcher to balance the two priorities, which are to protect the identity of participants and ensure the value and integrity of the participants data (Saunders, 2015). This means that the data collected from the study does not include identifying information such as names, addresses, identity numbers or any information that can be traced back to the individual responses of the participants identity. All participants were allocated pseudonyms and responses were coded to maintain confidentiality and anonymity during analysis and reporting. The data collected was stored in a device with a secure code and will only be accessed by the researcher. Hard copy transcripts as well completed consent forms will be stored in a locked cabinet at the University of Pretoria, Department of Psychology for ten years.

3. 8. 3 Emotional Risk

In accordance with the non-maleficence principle, a researcher is obligated to avoid harming research participants. In addition, it is required that the researcher refrains from worsening the conditions of participants if it cannot be improved (Barrow, 2016). Taking into consideration that there might be a high probability that participants will experience emotional discomfort reflecting on their lived experiences during the interview might be emotionally provoking/triggering. The topic of the current study was considered to be potentially triggering, therefore may bring about emotional discomfort. However, the researcher contained emotional discomfort and made arrangements to refer participants for psychological assistance, when necessary, which was available at no cost to them within the SAMHS.

3. 7 Reflexivity

Roulston (2010, p. 116) defined reflexivity in research as "the researcher's ability to be able to self-consciously refer to him or herself in relation to the production of knowledge about research topics; The subjective nature of qualitative research is recognised by establishing how one's identity and contextual positionality contribute to the construction of the research process and findings, this positionality can be explored using reflexivity" The



researcher used three types of reflexivity, namely personal reflexivity, interpersonal reflexivity and epistemological reflexivity.

3. 7. 1 Personal reflexivity

The researcher must be prepared to challenge their own assumptions during reflexivity (Willig, 2013). Especially when conducting qualitative research as the researcher forms an integral part of data collection. Reflexivity focuses more on the researcher than on the phenomenon under investigation (Reid et al., 2018). Therefore, it requires acknowledgement of the reality that the researcher is an active participant of the research process and have direct impact on the outcome of the study (Braun & Clarke, 2019). The researcher is a Black SANDF woman who has previously deployed to Sudan seven years ago and experienced a loss of a colleague and nursed severely injured colleagues following an attack during deployment, leading to adjusting difficulties such as isolation, flashbacks, and emotional volatility post-deployment for almost six weeks. This was taken into consideration when conducting interviews with the participants. Since the researcher has experienced the subject that is being investigated, the analysis of the study could potentially be influenced by her own experiences and underlying assumptions.

3. 7. 2 Interpersonal Reflexivity

Interpersonal reflexivity refers to the critical examination of the researcher's relationship with the participants and how the relationship can influence the process (Patnaik, 2013). This is important in the current study as the researcher works closely with some of the participant. The researcher thoroughly explained what was expected of the participants during the process of data collection.

3. 7. 3 Epistemological Reflexivity.

One of the critical functions of the epistemological reflexivity is that it allows to uncover underlying assumptions on which arguments and stances are founded on (Olmos-Vega, 2023). In addition, epistemological reflexivity necessitates that the researcher reflects on the assumptions (regarding the world and knowledge) they made during the research process and how it influences the overall study (Willig, 2013). My positionality regarding service women in the SANDF is that they are not aware of the implications pertaining to mental health following deployment-related trauma. Therefore, most of the service women do not seek psychological assistance after returning home from deployment, life goes on as usual. This may leave them vulnerable due to unresolved trauma.



3. 9 Conclusion

This chapter presented and discussed the methodology guiding the current study. A qualitative research approach, grounded in interpretative phenomenology analysis was used as the basis upon which this study was designed. In addition, the chapter further discussed the measures adhered to in order to ensure trustworthiness. Reflexivity was also presented as well as ethical considerations.

CHAPTER 4: FINDINGS

4. 1 Introduction

The aim of this study was to explore the lived experience of Black SANDF women post deployment-related trauma. The objectives guiding this aim were twofold: to explore the lived experiences of Black SANDF women; and to explore how they coped after experiencing deployment related trauma. This chapter focuses on the findings that were collected through semi structured interviews, which were recorded on audio then transcribed; Six participants were interviewed, interviews lasted between 25 minutes to 45 minutes for each participant. This chapter provides themes and sub themes that developed from the data, followed by chapter 5, which discusses the findings. All the participants were asked about their deployment-related trauma experiences and how they coped with it.

4. 2 Themes

Six main themes emerged from the data along with 26 sub-themes which are discussed in the following sections. The participants reflected on their traumatic experiences, their coping strategies, and challenges encountered during and post deployment; as well as positive lessons that emerged out their experiences. They also reflected on trauma symptoms they encountered during and after deployment and the support they received from their families during this time. The table below outlines the themes and subthemes of the current study.

Table 1:

Theme outline 1.

Main themes	Sub-themes





• Ethical changes faced by HCPs in deployment to a point of fearing consulting

4. 2. 1 Theme 1: PTSD symptoms

During the analysis of data themes pertaining trauma symptoms emerged, these include sound trigger, flashbacks sleep disturbance, isolation, and quick to anger.

4. 2. 1. 1 Sound trigger. Most of the participants indicated that they were triggered sound following trauma as it reminded him of the shootings during deployment. It was noted that banging sound and fireworks were more triggering as they were associated with explosives and rifles that were used in peacekeeping. See quotes below as evidence.

"OK, it was very challenging for me to adapt when I come back because a lot had changed there. You wake up anytime when you hear something. So maybe if the door bangs or somebody bang something then I will get scared." [Promise]

Ellesse shared the same sentiments: "Every time when you hear a banging sound, you think of the shootings...And when the kids play, they would bang the door and you kind of like snap out". [Ellesse]

From the experience of the participants, it seems like sounds that resembled those they experienced during their traumatic encounters evoked feelings of anger and fear in some.

4. 2. 1. 2 Flashbacks. Dimpho expressed how she experienced flashback following her traumatic encounter. "Whilst you sit normally when you are eating you are quiet and then there comes that person you helped with a lot of blood and scent and everything, people who died on the scene, you know, they, they come in your mind and all of a sudden, you're full you don't want to eat anything anymore. You start crying just sitting alone, thinking of the family of those people, thinking about the country that you are helping at the moment". [Dimpho, 28] Ellesse also shared her experience with flashback. "It's like your mind has not switched on that you are back home. So, if you hear a sound, you will just snap. Your mind is still at DRC, but your body is here. I was scared of the fireworks, especially in December". [Ellesse, 2]



What Dimpho and Ellesse described are flashbacks, detailing how they relived the traumatic events they had experienced. It is noted that the flashbacks occurred unprovoked with Dimpho in the form of visual images and smell, evoking intense emotions. While they were triggered by sound with Ellesse and elicited feelings of fear for sound that was similar to that of a rifle.

4. 2. 1. 3 Sleep disturbance. Out of the six participants, four stated that they struggled with interrupted sleep. As a result, some resorted to self-medicating to sleep peacefully. Dimpho mentioned that: *You know, at night when I struggled to sleep. I will take allergex and it will just (clicks finger) take me to where I want to go to sleep. [Dimpho, 3]*

Promise added that "so, it was hard to sleep after that it took me a while to eventually sleep normal". Ellesse not only encountered sleep disturbance but nightmares as well. "You won't sleep, you will have those nightmares". Mabontle also shared how she was not able to sleep peacefully and would be waked up by any sound: In the deployment area you are always on your toes. You can never sleep like a baby. Any sound that you hear wakes you up. So, I think the mind just get to be disrupted in sleep". [Mabontle, 1]

It was noted that some struggled during, and some struggled after deployment, while others struggled with sleep both during and after deployment. Both quality and quantity of sleep is negatively affected by trauma which was the case with some of the participants in the current study. On the other hand, disturbance in sleep is a normal response to trauma, meaning it was expected that the participants will present with sleep problems following their traumatic encounters.

4. 2. 1. 4 Isolation. Some of the participants indicated that they wanted to be alone as they felt irritable in the presence of other people, especially if they were around a lot of people for an extended period of time upon returning from deployment. The participants attributed this feeling to the fact they spent more time alone during deployment without family members or friends. These experiences can be seen in the following...

"So, those are the things that I encountered after coming back home, it took me a while to go out and see other people because when deployed, we were always inside the base......Yes. I was feeling like I'm suffocated when there was a lot of people around me". [Promise, 3]



I was very irritable; I didn't want to be in a space with my siblings and parents for long. I felt they're too much at times. I wanted my space because remember in deployment you mostly by yourself. You have your own room and space, so it was a bit too much on me. [Mabontle, 2]

4. 2. 1. 5 Quick to anger. Among other symptoms that were reported by the participants was that they were quick to anger post deployment related trauma, they were emotionally volatile, they were quick to respond in anger constantly even in minor issues. "The next thing you snap at them. The next thing you answer them roughly, you do something that they not used to. They will be surprised, and they will ask themselves a lot of questions". [Dimpho, 4]

Dimpho explains how she had changed in character following her experience. She mentions that she was this rough person meaning she had changed and became aggressive.

Mabontle also shared her experience: "We were irritable and annoyed throughout the speech and remember now that everyone is back from a long deployment, and they are dealing with a lot of things. So, I'd say that was the support we received on return...There were already behavior changes from the moment we landed. I remember in the very first week we had a medical repatriation. Somebody experienced mental health issues".

[Mabontle, 10]

From the findings or from the above quotes it appeared that there was a change in behavior even in the early stages of deployment. Antonette further shared; how minor things triggered her to a point of snapping. "I think they were very understanding because you find out the simple things though, they will trigger me like I'll just get angry, if I ask someone something or let me say I ask someone to go fetch something for me and then they take forever, I'll get angry and shout". [Antoinette, 3]

4. 2. 2 Theme 2: Coping Strategies

The participants recounted an array of coping strategies they used to cope following their traumatic experiences. For the purpose of this study the coping strategies were divided into two groups namely negative and positive coping strategies. While self-medicating and spending allowances comforting were categorised as negative coping strategies, prayer, venting, debrief, keeping fit/exercising as well as socialising were grouped as positive coping strategies. However, some themes seemed to overlap. It was noted that venting can be



classified as both a negative and positive coping strategy. In the current study participants reported that they used venting as a coping strategy to help mitigate emotional distress, therefore it was utilised as a positive coping strategy.

4. 2. 2. 1 Prayer. Religion plays a crucial role in the cultural and spiritual identity among South Africans, in addition religion goes beyond prayer, it is also seen as a form of support system that brings individuals, families and communities together. The participants engaged in various coping strategies to cope with the trauma, prayer was identified by three participants as means of coping. The participants indicated that prayer made them feel "better". Diteboho explained that after prayer, fellowship and reading, she was strengthened emotionally and spiritually. She further shared that her faith sustained her during times of difficulty and is an integral part of her life in general.

I think in in your career path, you quite understand this very clear that the person beliefs play a major role. You see, if you believe that this page can give support to you, it can do whatever you believe. So, I personally am a Christian and, on most occasion, I'll definitely pray to regain strength and visit my bible. [Diteboho, 2]. Promise added that: I'd just ask the ladies in the tent that we pray before we sleep. After praying, we felt better, and we will all go to sleep. [Promise, 1].

Dimpho indicated that she would pray for herself and her colleagues not to be injured or killed during an attack: You fear for your life. That is what we are going through during that time and then I'd pray for me and my crew not to be killed or injured. [Dimpho, 2]

It seems as though prayer became a beacon of hope to the participants, which not only helped them cope with trauma, but provided them with strength to cope with other deployment stressors as well.

4. 2. 2. 2 Psychological debriefing. Psychological debriefing is a psychological intervention provided early post a traumatic encounter (Tan et al., 2022). As previously mentioned, peacekeeping can be volatile, which requires that peacekeepers have a higher level of resilience. Providing debriefing to deployed SANDF members might benefit some of them, as it will offer them the opportunity to share in depth details of the traumatic incidents they witnessed and how they responded. In the current study two of the six participants opted for debriefing. The participants stated the following regarding debriefing: *Fortunately, because when we are deployed there are psychologists and social workers, usually after such*



experiences. We get called in the hall and go for counseling and that's what they did after the incident which was helpful.[Antonette, 2].

Ellesse added: 'There was a social worker and psychologist who came to give us counselling' [Ellesse, 2].

Though debrief is the main intervention utilised within the organisation to deal with deployment related trauma, it appeared to be used by few and not all participants considered it to be effective.

4. 2. 2. 3 Exercising. A vast amount of research has been documented regarding the benefits of exercise to reduce stress and enhance emotional well-being (Basso & Suzuki, 2017; Klaperski et al., 2019; Knapen et al., 2015). Some of the participants narrated that they utilised exercise as a coping strategy. For example, Antoinette mentioned that she used exercise and other activities such as reading and watching TV to "occupy" her mind to escape stressing. "Uhm luckily, I'm a very active person. I love keeping fit. I love reading just occupy my mind from a lot of stressors. So usually when we are back in the base, I keep fit, maybe go for a jog in the base, we were also doing taebo. So, you join taebo and then if you're not keeping fit out there you will be in the tent reading or catching up on some series. I guess that was my coping mechanism when I was out there. [Antonette, 2] When asked how she copes, Dimpho attested to Antonette's sentiment: maybe because I'm active and I'm a willing person [Dimpho, 4].

Engaging in exercise coupled with other activities for Antonette and Dimpho seemed to have a positive effect, as they alluded that it kept their mind 'occupied' from stressing and coped better during deployment.

4. 2. 2. 4 Venting. Some of the participants mentioned that they used venting as a means to cope following a traumatic encounter. They mentioned that after venting they felt "better", which can be seen in the following extracts. "Because sometimes if you are not right emotionally you can talk to someone then they can advise you and sometimes you go play games that you keep your mind busy". [Ellesse, 30]

"And going forward, I didn't consult the psychologist, but obviously 90% of my friends are psychologists, so I'd vent to them". [Mabontle, 4]



Venting appeared to be more preferred for Mabontle than using psychological services and to Ellesse, it was one of the options she made us of with other coping strategies in order to cope with negative emotions.

4. 2. 2. 5 Socialising. Another coping strategy reported by the participant was socialising. The extracts below reflect some of the participants' narratives about how they used to socialise to navigate challenges in the deployment area to cope with stress. Ellesse alluded that they not only worked during deployment, but the engaged in social activities. "You see, at least when we that side, it's not only about work. We'd entertain ourselves. We had braais and other events." [Ellesse, 2]

Dimpho mentioned how she engaged in activities such as ludo or chess to "delay" her from thinking about what had transpired, she also stated that while engaging in the activities they will talk about pleasant topics and not about the incident, in this manner she didn't get the time to think about the traumatic incident. I get involved in everything. If I see people playing something like a Ludo or chess, I go there. I get involved. If I see people sitting somewhere, I'll go there, and I become part of their conversation. You see will talk about anything; it delayed me from thinking about what happened. I didn't get a lot of time to think about those incidences. I always talk about other exciting events and everything. We talk about the weather, we talk about some events that we had, we talk about clothes and lifestyles and stuff like that. [Dimpho, 4]

- **4. 2. 2. 6 Self-medicating.** Two of the six participants reported that they self-medicated. One stated that she made use of allergex and the other made use of Gen-Payne which was previously described due to injury. Both participants indicated that they consumed medication so they could sleep peacefully after a traumatic incident. *I once got injured so I had Gen-Payne*. So, I'll take a pink tablet to go to sleep. [Promise, 1]. (Laughs) Not a lot, what helped me was allergex tablets... I'd go back to my medication, allergex takes me to sleep (laughs). [Dimpho, 4]
- **4. 2. 2. 7 Spending allowances comforting.** This sub theme encompassed the participants lived experiences regarding how their allowances helped them upon returning home. When asked how they coped with adjusting post deployment. They stated that allowances "helped". *It's much better because of sometimes you know that in your account there is a lot of money and you home with your family, so it helped.* [Ellesse, 2].



Honestly speaking I have. I have adjusted quite well when I came back and honestly, even our allowances helped as well. [Mabontle, 4]. It appears that for both participants, spending of allowances was more of a way to cope with negative feelings that resulted from their traumatic experiences.

4. 2. 3 Theme 3: Support System

This theme explores the identified support system of the participants during deployment. Sub themes that emerged from this theme were family support and support of colleagues on and off the battlefield.

4. 2. 3. 1 Family support. Three of the participants expressed how family support helped them cope deployment. Such support included keeping constant communication which helped decrease stress and worrying as well as logistically. The participants stated that they updated their family members on what was happening, however, they did not tell them everything as they thought it would bring about worrying and stressing to the family. Responses below as evidence.

My family was supportive they called almost every day, and that helped me not to worry a lot. Although I was not telling them everything, because they're civilian. However, engaging with them timeously helped me a lot. [Mabontle, 3]

You know, with my family, they supported me, and I supported them, and we agreed regarding deployment, so they understood though it was not easy, especially for my last born. [Diteboho, 3]

They were supportive in so many ways logistically and emotionally. You know so we kept in contact, they were always calling me to check up on me yeah. They would constantly check on me, asking if I'm fine, how are things going, they cared a lot. [Dimpho, 2]

4. 2. 3. 2 Support from colleagues on and off the battlefield. Most of the participants alluded that support from colleagues went beyond the call of duty. They likened the support to that of their family whilst away from family. For example, Ellesse explained that they had built a very strong bond overtime during deployment. *"They were very supportive, we were supporting each other, we were like a family because we are staying in*



the same place for a long time. So, we knew each other well and support each other since we were not in our original country". [Ellesse, 4]

Mabontle shares how the supportive the multidisciplinary team was: *They were such a supportive group. The social worker, the pharmacy, the nurse, the doctors, we were supportive of each other.* [Mabontle, 4]

Antonette further described the support as well as the synergy and teamwork that she had with her colleagues on and off duty during deployment. We were a team. We were doing things together, when they asked something to be done, everyone was participating. And then when we left, it was winter and when we arrived that side it was summer. So, our bodies were reacting. Some had swollen face, our tentmates will be there trying to help by giving you something to rub. So, there was real love you could see that ahh shame here I'm being loved, the support was there, like we cared for one another. We were supporting each other. [Antonette, 4]

The participants described what is referred to as unit cohesion, particularly horizontal cohesion that is bonds among peers, which seemed to be strong and was present on and off the battlefield.

4. 2. 4 Theme 4: Lessons from deployment

Despite the traumatic events encountered by the participants, it appears that the experience also brought about positive changes such as growth, gratitude, and financial discipline to some of them.

4. 2. 4. 1 Growth. Promise shared that she experienced personal and career growth: *I* learnt a lot. *I* learnt new things that were in the DOD that *I* didn't know about, other fighting corps, weapons and what it's like to be at war cause sometimes when they say the rebels are there, you know you must be prepared. *I* learnt how much strength *I* have. Yeah, there's a lot of things, I've learned to work under pressure. At some point in time our PERS officer had to come back home because she lost her parent, so I had to act as a PERS officer. So, there's a lot I learnt work wise. [**Promise**, **1**]

Diteboho also added that she learnt to work under adverse conditions, which lead to growth in her career and spiritual life. The lessons, I feel like I grew in the part of my career because I was able to render a good service, regardless of what was going on yeah, and I



learned to work under hard conditions, unbearable conditions, I must say. And yes, I grew more in my faith too. [Diteboho, 3]

Mabontle further shared that: I think that year for me, I feel like it was like a year of rain. So, I felt like myself as a psychologist, it was a season for me to leave my mark and make an impact. Which I believe I did and continued with my life. Now when I meet the people, I deployed with whether I'm going to another course or to do a workshop they still remember me. They still remember the work that I did. I feel proud for having done that irrespective of the circumstances, the challenges we encountered, especially when travelling to other bases.

[Mabontle, 4].

It is noted that though they were numerous challenges during deployment,

Mabontle experienced growth career wise and left a good impression as she was still recognised for the services she rendered.

4. 2. 4. 2 Gratitude. Some of the participants indicated that exposure to living conditions in deployment resulted in them being grateful for what they have and their country of origin. For example, Antonette mentioned that: But then I was like, I must be grateful, I have the bed, I have a house, I don't sleep hungry you know, life is not that bad, but yhoo that side it's really bad, from what I saw I have to be grateful for what I have no matter how little it is. [Antonette, 3]

Ellesse opined that although her country had its own shortcomings, but they were nowhere close to what she had witnessed in Mozambique, this is what she said: *In our country there's abuse, harassment, and all other horrible things, but we are better, we are blessed, here others are struggling and suffering more than us.* [Ellesse, 2]

Mabontle attested to Ellesse's sentiments: The country is always going through a lot and having just to be grateful of the little things that we have, you know you have a shelter over your head. [Mabontle, 2]

For Dimpho, exposure to such conditions not only lead to gratitude for life and relationships with people close to her, but also resulted in a change in lifestyle: You know, I was this person who's, like, living I can say a careless life, but since I have seen the life that side now, I'm being careful with everything that I do. I've been appreciative and I value everyone around me because the next thing you see someone the next moment the person is gone, gone.

[Dimpho, 7]



4. 2. 4. 3 Financial Discipline. This theme explored the participants lived experience regarding financial discipline following deployment. Ellesse mentioned that she deployed due to financial constraints, in turn she learned to be financially disciplined. However, if she did not have any financial problems, she would not have deployed: *I was struggling financially, so I had to go because of my financial problem, but if I didn't have a financial problem, I was not going to go. I'm disciplined when it comes to money.* [Ellesse, 2]

On the other hand, Dimpho observed how the locals survived with little to no money which in turn motivated her to prioritise her needs and be disciplined financially. *I can say, yeah, to be disciplined in a different way. For example, be disciplined financially. I used to use money to buy things that I don't really need sometimes, but it taught me to prioritise, to get my priorities right.* [Dimpho, 7]

It appears that due to their deployment trauma, their views changed towards life and finances. They reported to be more appreciative and grateful for the people in their lives, their country and life itself. What the participants described in the above extracts is the concept of posttraumatic growth (PTG). A concept coined by Tedeschi and Calhoun (1988) which defines positive transformations encountered by some individuals following trauma.

4. 2. 5 Theme 5: Challenges

This theme explored various challenges encountered by the participants during and post deployment. Five sub-themes emerged from this theme specifically lack support from the organisation during and post deployment, no leave, family challenges-integrating into family life upon return, attacks and ambushes by local community and witnessing locals suffer.

4. 2. 5. 1 Lack of support from the organisation during and post deployment.

Promise expressed her plight during deployment regarding the lack of support from the organisation. One of the challenges mentioned was the struggle to access basic necessities such as food. This is what she had to say: I'm not sure if this is going to be OK. Even our department, the DoD, it doesn't really support its military members who are outside. We struggled with a lot of things, we sometimes had a shortage of food, sometimes we just eat one thing for a long time, no network you can't call home. There was a lot of things. It was my first deployment, so that was my experience. It was hard to reach home. [Promise, 4]



The concern regarding the struggle with basic necessities was confirmed by Mabontle You know from the living quarters to other logistics; it was a challenge. The quarters are structured for males. [Mabontle, 7]

Mabontle further added that she was under the assumption that everything will go well however, but that was not the case. The situation was not what it had been portrayed as. She also alluded that accommodation was structured to suit males hence the challenge. Yeah, yeah, it's something else. During pre-deployment, they like selling you a diamond. They make it seem like there's a diamond and you need to go. When you leave you thinking things are going to go well. But now when you get that side, it's not anything like that diamond, they sold you.

On the other hand, Diteboho expressed her dissatisfaction with the organizations lack to provide emotional and psychological support. "There was not enough support. Actually, there was absolutely no support. You know, you would expect the organization when these incidents happen, they'd probably send the professionals' social workers and psychologists to come and check on us maybe twice. Even if it was once in a year, it would better, because even though I'm a social worker, I'm as good as a soldier, like our soldiers you know my life was in danger I must say. You see so I thought that it was going to be easy if they were sending people to come and you know, check on us.

[Diteboho, 2]

Diteboho asserted that she expected psychological support to be provided by the organisation following traumatic incident. Even though her job was to render the very service, she also needed it, as she was also exposed to traumatic incidents like the rest of the soldiers she was deployed with. She further opined then she was of the view that receiving psychological support as a healthcare practitioner from the organisation would be "better even if it was once". It was noted that health care workers mentioned a lack of psychological support, while those that were deployed to do groundwork such as patrols mentioned more of logistical challenges.

4. 2. 5. 2 No leave. Participants noted the challenge of prolonged deployment with no leave. The extracts below reflect on agreement from the participants regarding this sub theme.

And one other thing with our battalion, we were never given a chance to take leave and I personally I overstayed because I was there for full 15 months. [Diteboho, 3]



Yeah, it is quite a long time, I must say that in that 14 months we never even got a chance to go home. [Mabontle, 2]

4. 2. 5. 3 Family challenges-integrating into family life upon return. Adjusting after a long deployment can be challenging. To grasp the lived experiences of Black SANDF women after deployment-related trauma, this sub-theme focused on the family challenges they experienced upon returning home during integration. Both participants explained how they had to adapt to the new way in which their families were doing things. Promise indicated that *I had to start from scratch*, because when *I left*, they were not in the same level as *I came back*. They were older. And so, *I had to adapt to their situation at that time*.

Antonette added that: So, it was not that easy because when I returned, I got home the kids that I left that was six years now they were seven years. They can talk back. So, it was a bit challenging cause now when you want to do things this way, they were doing it their way, they have been doing it like this for the whole year. So, you have hold back and observe, see how they are doing things and then adjust or maybe you advise them. And ask OK, how about we do it like that, if they don't feel like agreeing with you, then you have to be OK and relax cause there will be clashes. [Antonette, 3]

Promise mentioned that it seemed like there were limits or boundaries when she was interacting with her daughter as she had grown up and was a changed person until her partner intervened. "To understand what I can do now, and what I can't do with my daughter. She grew up when I was not here. Then her father had to talk to her. But when I came back, she was like having attitude. And I had to adapt to her mood sometimes". [Promise, 3]

It appears that both participants felt that their children had grown while they were away on to for over a year. It is noted in the study that Promising and Ellesse experienced a similar family challenge specifically dealing with behavioral changes in children whilst integrating into family life post deployment.

4. 2. 5. 4 Witnessing locals suffer. This sub-theme emerged from the participants narrating their life experiences of witnessing the suffering of locals in the deployment area. Antoinette shared her experience: *Kids walking bare foot, carrying heavy things. And so, I looked at how they were they are living and then compared to the way we are living this side, or I am living this side. Yes, I don't have everything, and life is not perfect. [Antonette, 3].*



She compared the experience to the way of living in her own country. She voiced that the hardships she encountered in South Africa were nothing in comparison to the suffering of the locals. Ellesse described the dire conditions that engulf the locals: Let me maybe start with how it is in DRC. The people are suffering of poverty, it's too much, kids don't have parents, they're staying alone in small houses and then there's groups of rebels who threaten, kill, and burn their houses. [Ellesse, 1]

Diteboho indicated how traumatic it was to witness women and children in such living conditions. They ran into our base, and I was very traumatised to see how people in the DRC live. Witnessing women and children suffer was traumatic. [Diteboho, 2]

Mabontle shared how one can be ungrateful about the little things yet there are countries like the DRC that lived in extreme lack. *Honestly speaking, to see how people are going through a lot and you complain about drinking a black coffee, but somebody doesn't even have the tea to begin with.* [Mabontle, 2]

While exposure to the violation of the locals' human rights was traumatic, it seemed as though, it had allowed the participants to reflect on their own lives and living conditions back at home.

4. 2. 5. 5 Attacks by community or locals. Another challenge experienced by the participants was being attacked by the local community members and rebels. Mabontle narrated her experience of being attacked by the angry community members who believe that they were in their country for financial gain. when they travel by road.

"We were travelling in fear you know, when we passed through communities, they would throw stones and as I was travelling in an ambulance and it's a soft skin. So, you can imagine the anxiety there. You wondered whether you would make it...it was angry community members, they would be yelling we see you guys, you saying you are here to protect us, but we do not see any changes we still struggling...You know, and the rebels still came to attack us be like, what are you here for? You might as well just go back to your countries because we don't see impact since you came here or l you just came here to make money and go back. While we still in the same situation. We're losing our family, our loved ones. We're losing our things, we get displaced, we're losing our homes and everything we own. So, the community was angry then what they will do when you pass through, they would throw stones at you. A lot of trucks inventory, especially the driver's sides windows were broken. [Mabontle, 8]



When asked how often they encountered the attacks from the community, Mabontle mentioned that they were attacked a lot. It seemed that the presence of the peacekeepers was not helpful to the community as they mention that they were still going through atrocities of losing their family members and homes. Mabontle explained that they encountered such attacks numerous times especially.

4. 2. 6 Theme 6: Barriers to seeking mental health services.

This theme explored the participants' experience on why they could not seek psychological service during deployment following deployment-related traumatic encounter. Four sub-themes emerge from the participants sharing their experiences, specifically fear of changing medical category, shame of being perceived as not coping, debriefing not effective and ethical challenges faced by HCPs in deployment to a point of not consulting.

4. 2. 6. 1 Fear of changing in medical category. Both participants explained that they feared that consulting or seeking psychological help would result in a change with their medical category: "Remember, if I go see a psychologist at work or psychiatrist it affects your CHA and then you have to go through talking on whatever you saw." [**Promise, 2**]

Because I couldn't consult with the psychologist in the mission area, which I believe that she couldn't consult with me as well... I volunteered to deploy, I must say and as soon as I consult, ... I wouldn't want to be repatriated because when I was in the mission area, my medical category was affected, so I push on irrespective I had to develop a thick skin to be able to bounce back. [Diteboho, 2]

The above extracts reflect the reasons why some of the participants did not consult healthcare practitioner. Both participants were of the view that seeking psychological help would the results in a change in repatriation as well as medical category. Furthermore, a change in medical category may also result in slowing career progression as members cannot attend any promotional courses due to a change in medical category, this might create a barrier to seeking psychological help for some.

4. 2. 6. 2 Fear or shame of being perceived as not coping. Prior to deployment members of the SANDF are assessed for risk behavior and whether they can cope under adverse conditions during the Concurrent Health Assessment CHA assessments (Bester, 2022). In cases where members consult deemed as not coping during deployment, they are returned to their home unit. Some of the participants stated that they were ashamed of be



viewed as not coping, hence the decision not to seek psychological assistance. They had the following to say:

I might be seen as weak and that might affect my medical category and in as much as it's risky for one to be in DRC. [Diteboho]
Remember, if you talk sometimes, they might repatriate you, and say you are not coping you see. [Promise, 2]

4. 2. 6. 3 Debriefing not helpful. While some of the participants in this study found debriefing to be effective, this was contrary for Dimpho who opined that debriefing was ineffective as it was "just talking". She added that she still experienced flash backs, therefore, she preferred socializing and resorted to self-medicating to cope with trauma effects. She opined that: They will ask questions about, how we coping? If there's anything that comes into your mind more often or changes your life pattern, maybe some dreams, nightmares. But I don't think they really helped (laughs). [Dimpho, 2].

It appears that to Dimpho's the debriefing experience was just a conversation questioning what happened and how she is coping, which she feels was ineffective.

4. 2. 6. 4 Ethical challenges faced by HCPs in deployment. According to the ethical standards trust is an integral or component of the client professional relationship (Varkey, 2021). A compromise in upholding this ethical standard may result only to blurring the professional roles and boundaries between professional and personal relationships, in turn potentially lead to a breach of confidentiality. Diteboho, a healthcare practitioner expressed her concern regarding how the healthcare workers mobilised with the entire battalion whom they had to offer psychological service to. "You see, the other challenge that I wish can be addressed. It's we as professionals, it doesn't really make sense that we mobilise with the entire battalion, because that on its own it affects us. It affects our clients because remember these people that I'm deploying with them, they are all my clients. So, if I to move with them from one place to another, sleeping with them in the same tent, eating together, there comes a situation whereby my ethical profession can be jeopardised, I must say, in as much as I'm a social worker".... At the end of the day, I'm a human you see, I'm sleeping in the same tent with my clients when they get into the bus in the truck, I'm with them. So, when I get in the mission area. [Diteboho]



From the above quote it is evident that the client professional relationship may have been affected or were no professional boundaries and that may have compromised her objectivity and potentially lead to a violation of privacy and confidentiality. In addition, Diteboho also mentioned that she spent a lot of time with the members for and engaged in several activities with the clients even prior to going to the DRC.

"At the end of the day, I'm a human you see, I'm sleeping in the same tent with my clients when they get into the bus in the truck, I'm with them. So, when I get in the mission area. I'm expected to be this person, you know to be more of a social worker to them it was a struggle, but one had to push that to happen. It's not easy, I must say. Remember I'm staying and sharing the same tent with my clients. When I answer my calls, they are just here, when they crack jokes, I end up laughing".

It appears that Diteboho as a healthcare practitioner spent more time with her clients outside the consultation room than in the consultation room this might be the reason some of her clients may not have had trust in the services that she offered as a professional boundary may have been blurred.

4. 7 Conclusion

The themes that emerged from the findings of this study explored the deployment related experiences and coping strategies employed by the Black SANDF women. Theme 1, PTSD symptoms present the women's experiences with PTSD symptoms. This theme consisted of five themes, namely sound trigger, flashbacks, sleep disturbance, isolation, and emotional volatility/quick to anger. The second theme, coping strategies comprised of seven subthemes specifically, prayer, exercise as a coping strategy, venting, debrief, socialising, self-medicating, and comfort in spending allowances, it focused on how women coped with their traumatic experiences, particularly the negative and positive coping strategies the used to cope. Family support and support from colleagues on and off the battlefield were subthemes that emerged from the third them titled support system; this theme explored the support system the women in the SANDF they received post exposure to traumatic events in deployment. The fourth theme, deployment challenges included five subthemes namely, lack of support from the organisation during and after deployment. Theme five titled lessons from deployment focused on the positive aspects experienced by the women peacekeeper as a result of deployment, this theme included three subthemes namely, gratitude, growth and financial discipline. The sixth theme explored barriers in seeking psychological service



among the women following their traumatic encounter. Four subthemes were identified under this theme namely, fear of change of medical category, shame of being perceived as not coping or weak, debrief not helpful and ethical challenges leading to fear of consulting in HCP's. The way in which soldiers experience and cope with trauma experiences is different given the dynamics of the military environment.

Chapter 5: Discussion

5. 1 Introduction

This chapter presents and discusses findings of the study in relation to the literature review and existing literature. The two research questions that guided the discussion of the findings are: what are the lived experiences of deployment-related trauma of Black SANDF women? and what positive and negative coping strategies are employed by Black SANDF women to cope with deployment-related trauma? Limitations of the current study are presented in this study. In addition, recommendations regarding coping with deployment related trauma are presented as well as the conclusion of this study. The following are themes that emerged from the analysis: PTSD symptoms; coping strategies; support system; lessons from deployment; challenges; and barriers to seeking mental health services. The findings are discussed below under the six main themes.

5. 2 PTSD symptoms

Military members that have recently returned from deployment are at a greater risk of developing PTSD symptoms resulting from of exposure to traumatic events they have witnessed or experienced indirectly (Blias et al., 2021). These include flash backs, hypervigilance, and negative thoughts (APA, 2022). This has been the case with some of the participants in the current study. Participants reported five PTSD symptoms, namely sound trigger, flashbacks, sleep disturbance, isolation and emotional volatility. From the findings, it was observed that out of the six participants, one did not report any PTSD symptoms. Among the five participants who exhibited PTSD symptoms, each reported not more than three of these symptoms. These findings were compatible with Shi et al. (2017), where they found that Chinese health workers did not present with any PTSD symptoms twelve months following exposure to physical violence. In an article review, Kabaday-Şahin & Sevil (2023)



asserted that not all traumatic events result in PTSD. Another possibility may be that the participant might have been experiencing a delayed onset of PTSD symptoms (Solomon, 2020) during the time of the interview. A delayed onset of PTSD symptoms in military members is common (Stępka, 2018). In a study conducted by Fikretolgu and Liu (2012) to investigate a delayed onset of PTSD symptoms among military members post deployment, results showed that a few Canadian forces experienced a delayed onset of PTSD symptoms, which may be the case with the one participant in the current study. Several other studies have noted a delayed onset of PTSD symptoms among active military members as well as in veterans (Goodwin & Rona, 2013; McNally & Freuh, 2013; Richardson et al., 2016). The causes of a delayed onset of PTSD symptoms among soldiers has not received much attention (Anyaegbu, 2021). Bryant et al. (2013) discovered that ongoing stress might be the cause of delayed onset. However, this may be contrary to the current study as none of the participants indicated that they were currently experiencing stress, although they had challenges post deployment, they mentioned that they adjusted well.

Some participants reported that trauma symptoms were still present though not as severe as before when they had recently returned from deployment, others stated that symptoms eased during adjustment. This is evidence that symptoms differ from one soldier to another, in addition there are many factors that play a role in the development of PTSD (Polusny, 2013) in a study investigated the experiences of trauma survivors; the results illustrated that some trauma survivors experience PTSD symptoms for a short period whilst others presented with persistent PTSD symptoms. It was noted that those who reported ongoing symptoms, stated that they were still not considering seeking help in the current study.

Out of the 17 PTSD symptoms, the participants in this study, as stated above one presented with no symptoms, and the other five reported not more than three PTSD symptoms. Therefore, it may be concluded that they presented with very minimal symptoms of PTSD. The reason for this outcome may be due to the support structures, such as family support and support from colleagues on and off the battlefield, they had during and post deployment. Another possibility to account for this might be that they had a social worker and psychologist in the deployment area. Similar findings were noted by Heinecken and Ferreira (2012) among a cohort of SANDF soldiers deployed in DRC. However, very few of the participants mentioned that they made use of the psychological service at their disposal during the operation. These findings were in line with those of Seedat et al., (2002) and Sibanda et al. (2020). It was also noted in the current study that most participants described



how their symptoms manifested emotionally, cognitively and behaviorally but were not able to pinpoint exactly what the symptoms are, except for the healthcare practitioner. Heinecken and Wilén (2021) observed the same in South African female peacekeepers. The trend of fewer soldiers utilising psychological services especially post deployment and the inability to point out PTSD symptoms observed may be an indication for the organisation to review its intervention to combat PTSD.

5. 3 Coping strategies

The current study identified various coping strategies employed by Black SANDF women post deployment-related trauma. Lazarus and Folkman (1984) developed a theory addressing coping. In this study it was noted that Black SANDF women employed positive, negative and in other instances both types of strategies to cope. Negative coping refers to coping strategies that are linked to high levels of psychopathology symptoms, such coping involves the use of substance, avoidance, and lack of help seeking behavior (Compass et al., 2017). Participants mentioned that they utilised the following coping strategies: prayer, psychological debrief, venting, keeping fit, socialising, self-medicating and spending allowances. Lazarus and Folkman (1984) asserted that coping is influenced by the persons coping resources, meaning the manner in which two people handle stress may differ significantly. Coping strategies comprise behavioral and cognitive efforts to manage adverse situations (Algorani & Gupta, 2022).

According to the theoretical framework grounding this study (Lazarus & Folkman, 1984), stress results from the transaction between an individual and their surroundings; the level of demands and resources available for dealing with burden determine whether stress is faced or avoided. The theory assumes that an appraisal can be a primary and secondary perception of a stressor, and thereby an evaluation of potential effectiveness and consequences of coping behavior (Lazarus, 1993).

A primary appraisal occurs as the first step according to the theory, during which one evaluates whether a particular situation directly affects them (Lazarus, 2020). In order to determine whether or not stress will be caused by the problem at hand, the mind will weigh the significance of the issue. In the current study, the risk associated with exposure to traumatic events in soldiers during deployment include mental health problems, particularly PTSD, as well as difficulties in adjusting well post deployment (Waller et al., 2012). Therefore, the stress or problem is appraised as being potentially harmful (Folkman & Lazarus, 2013).



Secondary appraisal is a step that follows the primary appraisal step within the transactional model of stress and coping (Lazarus, 2020). The model assumes that during this phase, an individual attempts to manage the stress effectively to avoid negative outcomes. Individuals determines whether they have adequate resources to deal with the stress (Folkman & Lazarus, 2013). The participants could determine the psychological services such as psychological debriefing offered by the psychologist and social worker available to them during deployment. When the resources are perceived to be inadequate to deal with the problem, stress response is triggered (Lazarus, 2020).

Lazarus and Folkman (1984) identified two possible coping strategies: problem-based and emotional coping. Within the theoretical model of stress and coping, specific coping strategies are either understood as adaptive (positive) or maladaptive (negative). Positive coping occurs when individuals have the resources to deal with the problem in order to attain a positive outcome (Algorani & Gupta, 2021). For example, the use of psychological service resource to consult with the healthcare practitioners during psychological debriefing, partaking in social events, religious gatherings, and physical training activities by the participants to mitigate deployment stress. According to the transactional model of stress and coping, coping strategies can be classified as positive if they are able to effectively manage stress in a long term. For example, two of the participants shared that they engaged in exercise and other activities to deal with the aftermath of trauma. Their experiences were in accord with existing literature that discussed the benefits of exercise to reduce stress and enhance emotional well-being (Knappen et al., 2015; Blais, 2021). Similarly, these findings were consistent with that of Morgan et al. (2017) in a study conducted to examine the association between mental health and coping behaviors among a cohort of military members post deployment, results revealed that exercise was linked to fewer mental health symptoms.

Conversely Peltzer and Pengpid (2019) found that strenuous exercise was associated with PTSD symptoms among a youth cohort in a South African study. In addition, it has been discovered that exercise is effective in decreasing stress and anxiety among individuals who have been traumatised and enables individuals to redirect their emotions constructively and as a result gain resilience (Blais, 2021). Furthermore, studies have been proven that engaging in exercise produces endorphins a chemical produced by the brain that naturally alleviates pain and help cope with or reduces stress (Basso & Suzuki, 2017). A meta-analysis by Knappen et al. (2015) evidenced that exercise can improve mental health, coping strategies, quality of life and physical health among patients with depression. The positive impact of exercise on



mental and physical health has been well documented (Blais, 202; Knapen et al., 2015; Mahindra et al., 2023; Morgan, 2017). According to research exercises cause an alteration in the neurochemistry in the brain that promotes mental health (Panjwani et al., 2021). In addition, exercise has been shown to improve sleep, concentration, and energy (Knappen et al., 2015).

Another example is one of the participants who reported that religious activities gave her the "strength" to cope with deployment stress, two other participants also attested to the positive effects of prayer to cope with trauma during deployment. Religious coping strategies such as prayer have been found to decrease stress, improve health, quality of life as well as the ability to deal with adverse conditions (Mahamid & Bdier, 2021). These findings replicated those of Park et al. (2016) in a study that investigated the relationship between prayer as a coping strategy, perceptions of trauma and mental health symptoms among recently deployed U.S. soldiers; results evidenced that soldiers who used prayer as a coping strategy presented with better mental health outcomes in comparison to those who did not. The same phenomenon was observed by Alomari et al. (2021) among nurses who worked at the emergency department. Contrary to this study Morgan et al. (2017) found that service members during deployment who used prayer as a coping strategy and consulted a chaplain screened positive for depression, PTSD as well as for general anxiety disorder.

Socialising was among the positive coping strategies reported by the participants. According to research socialising impacts hormonal balance, it produces a hormone called oxytocin, which is responsible for alleviating anxiety levels enabling individuals to cope well with stress (Jones et al., 2017). These findings were in line with a qualitative study conducted by Lin et al. (2019) that investigated coping strategies used during COVID-19 by UK citizens; it was found that individuals who utilised socialising as a coping strategy coped well with COVID-19 stress.

Venting was reported as a coping strategy by some of the participants. They mentioned that they felt better after venting about the incidence. Venting allows individuals to release their frustrations or negative emotions, which results in them coping better.

According to Trouw et al. (2023) venting might be a double-edged sword, meaning venting can be both a positive and negative coping strategy; the authors argued that while letting out emotions make individuals feel or cope better, as people are prone to be sympathetic and reassuring. However, when venting is persistently used as a coping strategy it can adversely affect the other person's emotional state and strain the relationship, as they may gradually be less sympathetic and supportive as the time goes on (Trouw et al., 2023). Although venting in



the current study is reported as a positive coping strategy, I concur with the authors, given that the peacekeepers deploy for an extended period, therefore using venting might not be a viable coping strategy in a long run for the service women.

Negative coping happens when an individual realises that the demands exceed the resources, which manifests in various ways, such as the use of substances or avoidance of anything that reminds the individual of the traumatic occurrence (Ding et al., 2021). Maladaptive coping decreases the experience of stress for a short time but does not help or exacerbates the problem in a long run (Babicka-Wirkus et al., 2021). For example, one of the negative coping strategies reported to be utilised by two of the participants was self-medicating with prescription medication as a means to cope with sleep disturbance during deployment after a traumatic event. On the contrary, Fernando et al. (2017) found that student athletes self-medicated post injury to enhance performance. In the current study two of the participants reported that they achieved what they wanted which was a peaceful sleep after self-medicating. However self-medicating provided a temporary relief or solution to the problem which can potentially be harmful over time (Vong et al., 2022). Risk associated with self-medication include wrong self-diagnosis, adverse effects of the medication, resistance to particular medication in future, overdose or underdose, as well as running the risk of being dependent on the medication (Vong et al., 2022).

Another example was an uncommon negative coping strategy reported by the participants using allowances or shopping to escape negative emotions post deployment. The participants of the current study expressed how their spending of allowances made them feel "better" when asked how they coped with trauma symptoms upon returning home after deployment. The same phenomenon was observed by Mattock et al. (2012) among a cohort of veteran women post deployment. Similarly, Harrison et al. (2012) noted the same experience by hurricane Katrina survivors, who reported spending as a coping strategy following the natural disaster. Contrary to the findings of the current study, Ashley et al. (2016) found that trauma survivors had a tendency of over saving or acquiring following their traumatic encounter. Azam (2023), postulates that the act of spending money may be a trauma response for other individuals; the author further states that this kind of behaviour is mostly seen as a coping strategy by people who have had a traumatic encounter previously.

According to Sandberg (2019) spending can be used as a type of self-medicating that offers a brief distraction from the negative emotions and memory connected to a traumatic event. The shopping may have provided temporary escape from the negative feelings. However, later the



participants might have to deal with the very emotions that they were trying to escape from, as well as financial problems.

Soldiers returning from deployment usually encounter stress related to deployment as well as have difficulties in adjusting post deployment (De Soir, 2017). Social support from family, friends and home unit, and religious coping can assist in easing the stressors and adjust well (Polusny, 2013). Lack of a good support system might result in worsening of PTSD symptoms and other psychopathological mental health illnesses (Vogt, 2011). The model of stress and coping is valuable in that, it makes it possible to identify the outcomes of coping, and therefore be able to develop effective coping strategies.

As stated above, the service women employed both negative and positive coping strategies as well as a combination of both. Two negative and five positive coping strategies were employed by the women in order to cope with deployment-related stress. According to the theory of stress and coping, a barrier to accessing positive coping strategies, may lead to the use of negative coping strategies (Lazarus, 1984). This was noted in the current study, most of the participants who engaged in negative coping also reported barriers to seeking help. For example, the case of the two participants who were self-medicating, also encountered barriers when attempting to seek psychological help.

5. 4 Support systems

In the current study it was noted that the participants relied on their fellow peacekeepers for social or emotional support during deployment. They stated that support from their fellow soldiers played an important role as it provided emotional support. It is apparent that the participant experienced cohesion during deployment. Unit cohesion can be defined as bonds among soldiers or military members. There are two types, horizontal cohesion a bond among peers, and vertical cohesion which is a bond between leaders and subordinates (Castillo., 2014). Cohesion has been proven to be beneficial for military members overall mental and physical health (Thomassen et at., 2018). These findings were supported by Campbel-Sill et al. (2022) in a study where the authors investigated the buffering effects associated with two types of unit cohesion namely horizontal and vertical. The authors found that horizontal cohesion buffers against PTSD symptoms and depressive symptoms. This may be one of the reasons why the participants presented with fewer PTSD symptoms in the current study. However, none of the participants mentioned vertical cohesion during deployment. This might be indicative of poor subordinate leader relationship. From



these findings, it is evident that such support structures are built on trust as well as shared experiences and contributed significantly to assist women peacekeepers cope with stress during the peacekeeping operation.

Furthermore, it was noted that in the current study all participants relied on family support. It appeared that family support played a crucial role in the participant's capacity to deal with stress. Participants indicated that they valued the constant communication with family during deployment as it alleviated stress and worrying. Studies have shown that indeed family support is important for service members morale during deployment (Gerwitz et al., 2014; Heinecken & Wilén, 2021). Operational stressors such as separation from family can negatively impact both military members and their family. The findings replicated that of Harmse et al. (2021) in a study that explored the lived experiences of prolonged separation due to deployment or military training among SANDF officers. Welsh et al. (2015) examined the impact or effects of social support on emotional well-being among deployed military personnel, the authors found that higher levels of social support were linked to positive mental health outcomes regardless of the adverse negative deployment experiences. Family support is therefore an imperative resource for deployed military members (Delajah et al., 2016).

5. 5 Lessons from deployment

According to this study some of the participants stated that they gained valuable or vital lessons about life and themselves. Gratitude, growth, and financial discipline are some of these lessons. Some of the participants expressed their feelings of gratitude after their deployment experiences. They mostly mentioned that they had a great sense of appreciation for their loved ones and the support they received from them as well as life itself. Some expressed that they began to appreciate the small things that they took for granted prior to their experience.

They also spoke about how witnessing adverse conditions in deployment made them more appreciative of the comfort and stability they have back in South Africa, their home country. Among other lessons the present study found that growth was a recurring theme across participants. They explained how they learnt new skills that developed their career, spiritual life and were able to work under unbearable working conditions. Another lesson that was notable was an improvement in financial discipline reported by some of the participants. The participants alluded that while they were motivated by finances to deploy, the experience



heightened the importance of making sound financial decisions. These findings aligned with the concept of post-traumatic growth (PTG) by Tedeschi and Calhoun (1996), a concept that refers to the notion trauma survivors undergo psychological transformations following a traumatic encounter, growth gratitude are the most common changes witnessed. Several studies (Gallaway et al., 2011; Mark et al., 2018; Mitchel et al., 2013; Ogińska-Bulik ,2017) have noted the post-traumatic growth encountered by some soldiers or peacekeepers, implying or suggesting that traumatic experiences can result in positive personal transformations. This was echoed in various studies (Collier, 2016; Tedeschi & Calhoun, 2004; Tedeschi et al., 2018) that posited that exposures to trauma can bring about essence of gratitude and provide a new perspective on life or life priorities.

5. 6 Challenges

Deployment comes with its own set of challenges. The challenges experienced by the participants have been experienced previously within the same context as well as in forces across the globe, one of the challenges was the lack of support from the organisation during and post deployment. They mentioned how they lacked basic necessities. The findings were consistent with that of existing literature (Bruwer, 2021; Harmse et al., 2017; Heinecken, 2021; Mtshayisa & Letšosa, 2019) that found the SANDF to be lacking in terms of supporting its deployed members. Similarly, to a study conducted by Visagie et al. (2021) to compare the deployment experiences of the SANDF members during PKO's in Sudan and the DRC; the findings revealed that SANDF's peacekeepers deployed in both DRC and Sudan experienced a lack of support with basic necessities such as food, water, and accommodation from the organisation as a challenge. This indicates that the lack of support from the organisation is a long-standing challenge that requires urgent attention.

Some of the participants discussed their experiences regarding the lack of support from the organisation during and post deployment. The participants also noted a lack in providing emotional support to deployed health care workers. These findings collaborated with those of Gershon et al. (2016). It was noted that participants who deployed on the ground mentioned lack of support concerning logistics, while those deployed as medical staff expressed that they experienced a lack of emotional support from the organisation. Addler (2017) postulated that health care practitioners deployed in operational theatres face numerous demands that can cause burn out symptoms such as poor sleep patterns, ethical



dilemmas, workload as well as PTSD symptoms resulting from their work. In the present study, some participants mentioned some of the demands Adler (2017) referred to.

The participants also expressed the emotional strain of continuous service without leave during the deployment period. A possible explanation for the organisation might be its increased involvement in the PKO's. To fulfill the needs of the UN and SADC in Africa, the SANDF is required to deploy in PKO's as well as in other noncombatant operations, which increases the organisations involvement, thus increasing the risk of trauma among the SANDF deploying personnel (Harmse et al., 2022). The SANDF PKO's duration is approximately one year but could be extended. While the organisation may be obligated to fulfill its mandates to the UN and SADC, it is imperative that it considers the long-term adverse impact it may have on its members.

Among other challenges noted was family challenges during adjustment. Wegner et al. (2011) alluded that military members returning from extended deployments frequently anticipate going home, but the reality of homecoming may differ from what they envisioned. It is normal for soldiers to experience changes as well as their families, particularly for spouses and children left behind; in addition, the authors stated that when soldiers return home, they frequently find it difficult to integrate with their families, as the family dynamics might have changed during their absence. This can make them feel as though they are not needed, which creates problems for both the family and the returning soldier (Wegner et al., 2011). This was a reality for some of the participants who reported that they felt like they missed out on seeing some of the developments or milestones in their children's lives, as a result they also felt left behind. They further expressed how the changes in which their family managed their lives was challenging to navigate. Another challenge noted by the participants upon return was the stress pertaining to children's behaviors, which has been previously noted by several authors (Creech et al., 2015; Gerwitz & McMorris, 2014; Knobler et al., 2016; Stepka & Colhoun, 2016). Knobler et al. (2016) maintains that the parent-child relationship can be under strain upon integration, after the parent returns from deployment. These findings replicated a study by Creech et al. (2015) that examined the impact of separation due to deployment on parenting, the health outcomes, emotional and behavioral aspect of children. The authors found that children across all age groups presented with emotional and behavioral difficulties, as a result of deployment of parents (Creech et al., 2015). Similarly, this was echoed by Gerwitz and McMorris (2014) who explored family adjustment of deployed and non-deployed mothers among Afghanistan soldiers; results



showed that mothers experienced child functioning difficulties post deployment as well as other challenges.

To offer a possible explanation of the children's behaviour towards their parent upon return from deployment, Knobler et al. (2016) argued that children go through the experience akin to those noted in relationships of adults. Stepka and Colhoun (2016) discovered that the children's reactions differ across age groups and with parenting styles, younger children tend to cry, be clingy, and show avoidance or attention seeking behaviors; while the older children usually demonstrate being resentful, anxious, defiant and display various behavioural problems.

Witnessing locals suffer and attack by locals and rebels were among the challenges expressed by the participants. According to Mashishi (2013) the main factor that determines the PKO's success or failure, is the relationship among the peacekeepers and the local members of the community; the relationship between these parties is influenced by several factors such as attitude and behaviors of peacekeepers. The core purpose of deploying peacekeepers in the DRC is to provide humanitarian assistance, which includes protecting and safeguarding the local population (Dervishi et al., 2015). It was noted in the findings of the present study that the participants reported being constantly attacked by the local community members who believed that the peacekeepers were in their country for mere financial gain. According to the participants locals lamented that they were still experiencing dire conditions even in their presence. Therefore, the peacekeepers were perceived as not having interest in the problems encountered by the local community. These findings were in accord with Van lees (2020) in a study that explored the experiences of soldiers and locals; the soldiers reported to have experienced clashes with the locals as they were perceived as the elites in their community whilst they suffered. As stated above one of the core mandates of peacekeepers is to provide humanitarian support to locals, however the hostile relationship noted between the peacekeepers and the locals can hinder the fulfillment of the core mandate. Austere (2019) argued that in order to gain cooperation from locals, it is imperative that locals be the main actors or take the lead in peacekeeping as they are familiar with the issues in their land.

5. 7 Barriers to seeking mental health services.

Some of the participants reported barriers to seeking medical services such as fear of change in medical category, shame of being perceived as not coping and some stated that



psychological debriefing was ineffective. Some of the participants stated that they did not seek medical help as they were ashamed of being perceived as not coping or being perceived as weak. One of the barriers of seeking psychological service or mental health services mentioned by some of the participants was that they feared that they would be repatriated, and their medical category would change. These findings were in accord with that of Cornish et al. (2014) which found that women peacekeepers feared seeking psychological services following a traumatic deployment encounter because they feared it will affect their future promotions as well as future deployments. Similarly, Ntantamala and Adams (2022) investigated the correlates of PTSD among emergency care workers and barriers in seeking psychological care services. Barriers that were identified by the emergency care workers were that it would impact their career negatively. To better understand why the members feared the change in medical category, we briefly look at the process of the Concurrent Health Assessment in the organisation (CHA). SANDF personnel are required to undergo a medical examination CHA every two years. The CHA assessment is classified into three color codes, green which denotes medically fit, yellow denotes temporary medically unfit, while red means medically unfit. According to the policy of the organisation personnel members who are yellow or red medically cannot deploy or attend any functional courses or be promoted. From the policy put in place, it is apparent that change in the medical category may delay career progression or have a negative impact on career hence the fear of consulting which may result in a change in medical category.

Participants also mentioned that they did not seek help as they were ashamed of being perceived as "weak". This barrier of being perceived as "weak" has been observed across several forces globally (Gould et al., 2010; Langstone et al., 2007; Williamson et al., 2019). Drawing from the studies, it appears that being "perceived as weak" is not a barrier in the SANDF only or in a few militaries but across different militaries globally. This may indicate that military culture has an impact on how soldiers perceive seeking help. The military culture emphasises physical strength, resilience, stoic, discipline, honesty, courage, loyalty and perseverance (Michalopoulou et al., 2017). This culture is deeply embedded in soldiers as soon as they enter the system; a well-accepted element of military culture is to "toughen up", a deviation from this element may be perceived as a weakness in the military environment (Ganza et al., 2021). Though the military has embraced the use of military psychologists, the military culture by default does not encourage a conducive environment to seek psychological service (Gibson et al., 2014).



Belton (2016) examined soldiers' attitudes and perceptions towards psychological debriefing, combat exposure, and deployment stressors and the severity of PTSD symptoms among recently deployed soldiers; results show that soldiers who engage in psychological debriefing demonstrated a positive attitude towards the intervention. This was contrary to the findings of the current study as some of the participants who were engaged in psychological debriefing found it to be ineffective and showed a negative attitude towards it. Some of the participants described it as "just talking and does nothing to reduce or remove the trauma".

Debriefing is a main intervention used to combat PTSD after trauma exposure during deployment as well as after deployment during demobilisation (Dhladhla & Van Dyk, 2009). However, some of the participants found the intervention to be ineffective. These findings aligned with that of Vinguard et al. (2022) in a meta-analysis study conducted to examine the effectiveness of psychological debriefing among a cohort of soldiers and emergency workers. The debate regarding the effectivity of debriefing has been ongoing for decades. While other authors debated that it is not possible for a single session to prevent trauma symptoms, others suggested that it cannot be used solely as an intervention but can be used in conjunction with other early interventions (Resick, 2016; Richardson et al., 2016; Sareen, 2014). Tan et al. (2022) conducted a systematic review of literature to assess the effectiveness of debriefing and cohesion training on PTSD symptoms and common mental disorders in military members and emergency service workers. The authors found limited evidence regarding the psychological intervention's effectiveness or ability to reduce both PTSD symptoms and common mental disorders. However, results showed that participants who received a combination of both psychological debrief and cohesion training presented with significantly lower symptoms of PTSD and common mental health disorders, but no effects were found when psychological debriefing was given on its own (Tan et al., 2022). This was echoed by Hawker et al. (2010) in the meta-analysis that demonstrated that debriefing has limited efficacy. On the other hand, Tamarack et al. (2020) contended that debriefing may have been dismissed too quickly; the authors argued that the studies on the effectiveness of debriefing did not adhere to the procedures outlined in the Cochrane review article. In the current study some participants described debriefing as just "talking" which did not reduce their trauma symptoms, while others found it to be effective. It can be concluded that findings of the current study are inconclusive with regards to the effectiveness of psychological debriefing. However, it was noticed that most participants preferred other interventions than psychological debriefing which is the main intervention used during deployment.



5. 6 Recommendations

It will be beneficial if future research studies investigate the long-term effectiveness of interventions in the Defence Force such as debriefing. In addition, conducting quantitative studies to examine the statistics of soldiers with PTSD. Despite the traumatic encounters experienced by the participants, they still gained valuable lessons in deployment and experienced post-traumatic growth period, therefore exploring the concept of PTG across all arms of service might be beneficial. Lastly, exploring the impact of deployment on families of deployed members may help develop post deployment programmes to assist soldiers reintegrate well upon return from deployment.

5. 7 Limitations

The current study was not without limitations. Firstly, the sample consisted of Black SANDF women from Gauteng province and is therefore not a representative of all SANDF Black women. Secondly, there was a dearth in literature studies conducted within the SANDF, most of the literature used in the study was that of global military forces. Thirdly, some of the participants were not fully at liberty to express themselves with the concern that they might be traceable. However, I assured them of confidentiality and anonymity. This could potentially limit the study as the method of IPA aims to explore lived experiences in depth.

5.8 Conclusion

Peacekeeping missions require soldiers to be prepared emotionally, physically, and psychologically. Even though peacekeepers receive the same training pre and post deployment, they however cope differently with traumatic encounters and deployment challenges. The aftermath of trauma affects the family members of the peacekeepers as well. In addition, there are a plethora of factors that influence how soldiers cope during and after deployment. Furthermore, traumatic experiences not only result in traumatic responses but in positive transformations as well. The current study explored how Black SANDF women cope with deployment related trauma and how they navigate challenges during and after deployment.

The primary aim of this study was to explore the lived experiences and coping strategies employed by Black SANDF women following deployment-related trauma. The study was guided by the theory of stress and coping by Lazarus and Folkman (1984).



Participants of the study were six in total, who were recruited through non-purposive sampling. Data was gathered through semi structured interviews to explore the lived experiences of participants. The findings of the current study highlighted how SANDF women cope with deployment related trauma. Both negative and positive experiences of deployment, stresses experienced during deployment as well as problems associated with adjustment upon return from deployment were reflected on by the participants.



References

- Adler, A., Bliese, P., & Castro, C. (2011). Deployment psychology: Evidence-based strategies to promote mental health in the military. https://doi.org/10.1037/12300-000
- Adler, A. B., Adrian, A. L., Hemphill, M., Scaro, N. H., Sipos, M. L., & Thomas, J. L. (2017). Professional Stress and Burnout in U.S. Military Medical Personnel Deployed to Afghanistan. *Military medicine*, *182*(3-4), e1669-e1676. https://doi.org/10.7205/milmed-d-16-00154
- Adler, A. B., Bliese, P. D., & Castro, C. A. (2011). *Deployment psychology: Evidence-based strategies to promote mental health in the military* (Vol. 12). American Psychological Association Washington, DC.
- Adler, A. B., Bliese, P. D., McGurk, D., Hoge, C. W., & Castro, C. A. (2009). Battlemind debriefing and battlemind training as early interventions with soldiers returning from iraq: Randomization by platoon. *J Consult Clin Psychol*, 77(5), 928-940. https://doi.org/10.1037/a0016877
- Agha, S. (2021). Mental well-being and association of the four factors coping structure model: A perspective of people living in lockdown during COVID-19. *Ethics*, *Medicine and Public Health*, *16*, 100605. https://doi.org/https://doi.org/10.1016/j.jemep.2020.100605
- Alase, A. (2017). The Interpretative Phenomenological Analysis (IPA): A Guide to a Good Qualitative Research Approach [Interpretative phenomenology analysis, IPA, qualitative research method approach]. 2017, 5(2), 11. https://doi.org/10.7575/aiac.ijels.v.5n.2p.9
- Alchin, A., Gouws, A., & Heinecken, L. (2018). Making a difference in peacekeeping operations: Voices of South African women peacekeepers. *African Security Review*, 27(1), 1-19. https://doi.org/10.1080/10246029.2017.1406381
- Algorani, E., & Gupta, V. (2021). Coping Mechanisms. [Updated 2021 May 3]. *StatPearls* [Internet]. Treasure Island (FL): StatPearls Publishing.
- Algorani, E., & Gupta, V. (2022). Coping mechanisms. National Library of Medicine.

 National Center for Biotechnology Information. https://www.ncbi.nlm.nih.gov/books/NBK559031.
- Alsaigh, R., & Coyne, I. (2021). Doing a Hermeneutic Phenomenology Research
 Underpinned by Gadamer's Philosophy: A Framework to Facilitate Data Analysis.



- International Journal of Qualitative Methods, 20, 1-10. https://doi.org/10.1177/16094069211047820
- Álvares, R. S. R., Mendonça-de-Souza, A. C. F., Duarte, A. F. A., Gameiro, T. M., Fischer, N. L., Souza, W. F., Coutinho, E. d. S. F., Figueira, I., Volchan, E., & Souza, G. G. L. (2020). Exposure to stressful events during a peacekeeping mission may have a price: The impact on trait of negative and positive affect and mental health. *Journal of Health Psychology*, 25(9), 1285-1291. https://doi.org/10.1177/1359105317753715
- Anyaegbu, R. D., Nwatu, A. C., & Omeje, O. F. (2021). Social Support and Resilience

 Predicting Post-Traumatic Stress Disorder among Returned Soldiers from Insurgency

 Areas in Nigeria. *ESUT JOURNAL OF SOCIAL SCIENCES*, 6(3).

 https://esutjss.com/index.php/ESUTJSS/article/view/80
- American Psychological Association [APA]. (2022). *Trauma*. https://www.apa.org. Retrieved July 11, 2022, from https://www.apa.org/topics/trauma.
- Arancibia, M., Leyton, F., Morán, J., Muga, A., Ríos, U., Sepúlveda, E., & Vallejo-Correa, V. (2022). Phycological debriefing in acute traumatic events: Evidence synthesis. *Medwave*, 22(1), e8517. https://doi.org/10.5867/medwave.2022.01.002538 (Debriefing psicológico en eventos traumáticos agudos: síntesis de la evidencia secundaria.)
- Armenta, R. F., Rush, T., Leardmann, C. A., Millegan, J., Cooper, A., & Hoge, C. W. (2018). Factors associated with persistent posttraumatic stress disorder among U.S. military service members and veterans. *BMC Psychiatry*, *18*(1). https://doi.org/10.1186/s12888-018-1590-5
- Arthur, J., Walker, D., & Thoma, S. (2018). Soldiers of Character.
- Astill Wright, L., Sijbrandij, M., Sinnerton, R., Lewis, C., Roberts, N. P., & Bisson, J. I. (2019). Pharmacological prevention and early treatment of post-traumatic stress disorder and acute stress disorder: a systematic review and meta-analysis.

 Translational Psychiatry, 9(1). https://doi.org/10.1038/s41398-019-0673-5
- Atkinson, W. (2008). Not all that was Solid has Melted into Air (or Liquid): A Critique of Bauman on Individualization and Class in Liquid Modernity. *The Sociological Review*, *56*(1), 1-17. https://doi.org/10.1111/j.1467-954X.2008.00774.x
- Babbie, E. R. (2013). *The practice of social research* (13th, student ed.). Wadsworth Cengage Learning.



- Babicka-Wirkus, A., Wirkus, L., Stasiak, K., & Kozłowski, P. (2021). University students' strategies of coping with stress during the coronavirus pandemic: Data from Poland. *PLoS One*, *16*(7), e0255041. https://doi.org/10.1371/journal.pone.0255041
- Badampudi, D., Fotrousi, F., Cartaxo, B., & Usman, M. (2022). Reporting Consent, Anonymity and Confidentiality Procedures Adopted in Empirical Studies Using Human Participants. *e-Informatica Software Engineering Journal*, *16*(1), 1-24. https://doi.org/10.37190/e-inf220109
- Balderrama-Durbin, C., Erbes, C. R., Polusny, M. A., & Vogt, D. (2018). Psychometric evaluation of a measure of intimate partner communication during deployment. *J Fam Psychol*, 32(1), 31-41. https://doi.org/10.1037/fam0000382
- Banks, M. (2018). Using visual data in qualitative research. *Using visual data in qualitative research*, 1-192.
- Barrow, A. (2016). Implementing Security Council Resolution 1325 in Asia: The role of national action plans on women, peace and security. In *Gender, Violence and the State in Asia* (pp. 84-100). Routledge.
- Bartone, P. T., & Homish, G. G. (2020). Influence of hardiness, avoidance coping, and combat exposure on depression in returning war veterans: A moderated-mediation study. *Journal of Affective Disorders*, 265, 511-518. https://doi.org/https://doi.org/10.1016/j.jad.2020.01.127
- Basu, S. (2018). The women in blue helmets: gender, policing, and the UN's first all-female peacekeeping unit; Equal opportunity peacekeeping: women, peace, and security in post-conflict states. *International Affairs*, *94*(3), 663-665. https://doi.org/10.1093/ia/iiy081
- Bauman, Z. (1976). Towards a Critical Sociology: An Essay on Commonsense and Emancipation. Routledge & K. Paul. https://books.google.co.za/books?id=WjGyAAAAIAAJ
- Bell, J., & Waters, S. (2014). *Doing Your Research Project: A Guide For First-Time Researchers*. McGraw-Hill Education. https://books.google.co.za/books?id=vZ2RoAEACAAJ
- Belton, J. K. (2016). Soldiers' Debriefing Attitudes, Combat Exposure, and Deployment Stressors on Post-deployment Mental Health Symptoms Catholic University of America].



- Bertolozzi, F. (2010). women-blue-helmet-integration-women-and-gender-issues-un-peacekeeping-missions. https://reliefweb.int/report/world/women-blue-helmet-integration-women-and-gender-issues-un-peacekeeping-missions>
- Biggs, A., Brough, P., & Drummond, S. (2017). Lazarus and Folkman's Psychological Stress and Coping Theory. In *The Handbook of Stress and Health* (pp. 349-364). https://doi.org/https://doi.org/10.1002/9781118993811.ch21
- Blais, R. K., Tirone, V., Orlowska, D., Lofgreen, A., Klassen, B., Held, P., Stevens, N., & Zalta, A. K. (2021). Self-reported PTSD symptoms and social support in U.S. military service members and veterans: a meta-analysis. *Eur J Psychotraumatol*, *12*(1), 1851078. https://doi.org/10.1080/20008198.2020.1851078
- Bleckner, J. (2013). From Rhetoric to Reality: A Pragmatic Analysis of the Integration of Women into UN Peacekeeping Operations. *Journal of International Peacekeeping*, 17(3-4), 337-360. https://doi.org/https://doi.org/10.1163/18754112-1704009
- Bode, I., & Karlsrud, J. (2019). Implementation in practice: The use of force to protect civilians in United Nations peacekeeping. *European Journal of International Relations*, 25(2), 458-485. https://doi.org/10.1177/1354066118796540
- Braun, V., & Clarke, V. (2019). Reflecting on reflexive thematic analysis. *Qualitative Research in Sport, Exercise and Health*, 11(4), 589-597. https://doi.org/10.1080/2159676X.2019.1628806
- Brenner, L. A., Bahraini, N., Homaifar, B. Y., Monteith, L. L., Nagamoto, H., Dorsey-Holliman, B., & Forster, J. E. (2015). Executive functioning and suicidal behavior among veterans with and without a history of traumatic brain injury. *Arch Phys Med Rehabil*, 96(8), 1411-1418. https://doi.org/10.1016/j.apmr.2015.04.010
- Britt, T. W., Adler, A. B., Sawhney, G., & Bliese, P. D. (2017). Coping Strategies as Moderators of the Association Between Combat Exposure and Posttraumatic Stress Disorder Symptoms. *Journal of Traumatic Stress*, *30*(5), 491-501. https://doi.org/https://doi.org/10.1002/jts.22221
- Brooks, S. K., Dunn, R., Amlôt, R., Rubin, G. J., & Greenberg, N. (2017). Social and occupational factors associated with psychological wellbeing among occupational groups affected by disaster: a systematic review. *Journal of Mental Health*, 26(4), 373-384. https://doi.org/10.1080/09638237.2017.1294732
- Brown, B. A., Goodman, F. R., Disabato, D. J., Kashdan, T. B., Armeli, S., & Tennen, H. (2021). Does negative emotion differentiation influence how people choose to



- regulate their distress after stressful events? A four-year daily diary study. *Emotion*, 21(5), 1000-1012. https://doi.org/10.1037/emo0000969
- Brown M. P.H, M. A. E. K., Guthrie Ph.D, K. M., Stange, M., & Creech, S. (2021). "A Woman in A Man's World": A Pilot Qualitative Study of Challenges Faced by Women Veterans During and After Deployment. *Journal of Trauma & Dissociation*, 22(2), 202-219. https://doi.org/10.1080/15299732.2020.1869068
- Brunet, A., Monson, E., Liu, A., & Fikretoglu, D. (2015). Trauma Exposure and Posttraumatic Stress Disorder in the Canadian Military. *The Canadian Journal of Psychiatry*, 60(11), 488-496. https://doi.org/10.1177/070674371506001104
- Bruwer, N., & Dyk, G. A. J. V. (2005). The South African peacekeeping experience: a comparative analysis. *SA Journal of Industrial Psychology*, *31*(2), 30-39. https://doi.org/doi:10.10520/EJC89047
- Bryant, R. A. (2021). A critical review of mechanisms of adaptation to trauma: Implications for early interventions for posttraumatic stress disorder. *Clinical Psychology Review*, 85, 101981. https://doi.org/https://doi.org/10.1016/j.cpr.2021.101981
- Burkhart, L., & Hogan, N. (2015). Being a Female Veteran: A Grounded Theory of Coping With Transitions. *Soc Work Ment Health*, *13*(2), 108-127. https://doi.org/10.1080/15332985.2013.870102
- Campbell-Sills, L., Flynn, P. J., Choi, K. W., Ng, T. H. H., Aliaga, P. A., Broshek, C., Jain, S., Kessler, R. C., Stein, M. B., Ursano, R. J., & Bliese, P. D. (2022). Unit cohesion during deployment and post-deployment mental health: is cohesion an individual- or unit-level buffer for combat-exposed soldiers? *Psychol Med*, 52(1), 121-131. https://doi.org/10.1017/s0033291720001786
- Castillo, J. J., & Castillo, J. J. (2014). 16Cohesion Theory: Explaining a Military's Will to Fight. In *Endurance and War: The National Sources of Military Cohesion* (pp. 0). Stanford University Press. https://doi.org/10.11126/stanford/9780804789103.003.0002
- Castro, C. A., Adler, A. B., McGurk, D., & Bliese, P. D. (2012). Mental health training with soldiers four months after returning from iraq: Randomization by platoon. *Journal of Traumatic Stress*, 25(4), 376-383. https://doi.org/https://doi.org/10.1002/jts.21721
- Castro, J., Soares, M. J., Pereira, A. T., & Macedo, A. (2017). Perfectionism and negative/positive affect associations: the role of cognitive emotion regulation and perceived distress/coping. *Trends in Psychiatry and Psychotherapy*, *39*(2), 77-87. https://doi.org/10.1590/2237-6089-2016-0042



- Coetzee, R. H., Simpson, R., & Greenberg, N. (2010). Detecting Post-Deployment Mental Health Problems in Primary Care. *Journal of the Royal Army Medical Corps*, *156*(3), 196-199. https://doi.org/10.1136/jramc-156-03-16
- Compas, B. E., Jaser, S. S., Bettis, A. H., Watson, K. H., Gruhn, M. A., Dunbar, J. P., Williams, E., & Thigpen, J. C. (2017). Coping, emotion regulation, and psychopathology in childhood and adolescence: A meta-analysis and narrative review. *Psychol Bull*, *143*(9), 939-991. https://doi.org/10.1037/bul0000110
- Creech, S. K., Hadley, W., & Borsari, B. (2014). The Impact of Military Deployment and Reintegration on Children and Parenting: A Systematic Review. *Prof Psychol Res Pr*, 45(6), 452-464. https://doi.org/10.1037/a0035055
- Dahlberg, K., & Dahlberg, H. (2004). Description vs. interpretation a new understanding of an old dilemma in human science research. *Nursing philosophy : an international journal for healthcare professionals*, *5*, 268-273. https://doi.org/10.1111/j.1466-769X.2004.00180.x
- Danieli, Y. (2018). Sharing the Front Line and the Back Hills: International Protectors and Providers-Peacekeepers, Humanitarian Aid Workers and the Media in the Midst of Crisis. Routledge.
- DeJonckheere, M., & Vaughn, L. M. (2019). Semistructured interviewing in primary care research: a balance of relationship and rigour. *Fam Med Community Health*, 7(2), e000057. https://doi.org/10.1136/fmch-2018-000057
- Delahaij, R., Kamphuis, W., & van den Berg, C. E. (2016). Keeping engaged during deployment: The interplay between self-efficacy, family support, and threat exposure. *Military Psychology*, 28(2), 78-88. https://doi.org/10.1037/mil0000098
- Denzin, N. K., & Lincoln, Y. S. (2011). *The SAGE Handbook of Qualitative Research*. SAGE Publications. https://books.google.co.za/books?id=qEiC-_ELYgIC
- Dervishi, E., Ibrahimi, S., & Elshani, H. (2015). Post-traumatic stress disorder in militaries of peacekeeping missions in areas of wars and conflicts. *Academic journal of interdisciplinary studies*, *4*(2 S2), 11-14. http://dx.doi.org/10.5901/ajis.2015.v4n2s2p11
- Dhladhla, T., & Van Dyk, G. (2016). PTSD management in military forces in Africa. *Military Psychology for Africa*, 197-214.



- Dhladhla, T. J., & van Dyk, G. A. J. (2009). Management of Post-Traumatic Stress Disorder in the Military: An Intervention Model. *Journal of Psychology in Africa*, 19(4), 591-595. https://doi.org/10.1080/14330237.2009.10820334
- Di Razza, N. (2020). The accountability system for the protection of civilians in UN peacekeeping. *New York: International Peace Institute*.
- Di Salvatore, J., & Ruggeri, A. (2017). Effectiveness of Peacekeeping Operations. In: Oxford University Press.
- Ding, Y., Fu, X., Liu, R., Hwang, J., Hong, W., & Wang, J. (2021). The Impact of Different Coping Styles on Psychological Distress during the COVID-19: The Mediating Role of Perceived Stress. *Int J Environ Res Public Health*, *18*(20). https://doi.org/10.3390/ijerph182010947
- Dirkzwager, A. J. E., Bramsen, I., Adèr, H., & van der Ploeg, H. M. (2005). Secondary

 Traumatization in Partners and Parents of Dutch Peacekeeping Soldiers. *Journal of Family Psychology*, *19*(2), 217-226. https://doi.org/10.1037/0893-3200.19.2.217
- Doody, C. B., Egan, J., Bogue, J., & Sarma, K. M. (2022). Military personnels' experience of deployment: An exploration of psychological trauma, protective influences, and resilience. *Psychological Trauma: Theory, Research, Practice, and Policy*, *14*(4), 545-557. https://doi.org/10.1037/tra0001114
- Elnitsky, C. A., Fisher, M. P., & Blevins, C. L. (2017). Military Service Member and Veteran Reintegration: A Conceptual Analysis, Unified Definition, and Key Domains. *Front Psychol*, 8, 369. https://doi.org/10.3389/fpsyg.2017.00369
- Erciyes, E. (2020). Paradigms of Inquiry in the Qualitative Research. *European Scientific Journal ESJ*, 16(7). https://doi.org/10.19044/esj.2020.v16n7p181
- Fernando, A. D. A., Bandara, L. M. H., Bandara, H., Pilapitiya, S., & de Silva, A. (2017). A descriptive study of self-medication practices among Sri Lankan national level athletes. *BMC Res Notes*, *10*(1), 257. https://doi.org/10.1186/s13104-017-2579-8
- Feuer, B. S. (2021). First Responder Peer Support: An Evidence-Informed Approach. *Journal of Police and Criminal Psychology*, *36*(3), 365-371. https://doi.org/10.1007/s11896-020-09420-z
- Fikretoglu, D., & Liu, A. (2012). Prevalence, correlates, and clinical features of delayed-onset posttraumatic stress disorder in a nationally representative military sample. *Soc Psychiatry Psychiatr Epidemiol*, 47(8), 1359-1366. https://doi.org/10.1007/s00127-011-0444-y



- Flood, A., & Keegan, R. J. (2022). Cognitive Resilience to Psychological Stress in Military Personnel. *Frontiers in Psychology*, *13*. https://doi.org/10.3389/fpsyg.2022.809003
- Folkman, S., & Lazarus, R. S. (2013). Stress and coping theory applied to the investigation of mass industrial psychogenic illness. In *Mass psychogenic illness* (pp. 237-255). Routledge.
- Frank, C., Zamorski, M. A., Lee, J. E. C., & Colman, I. (2018). Deployment-related trauma and post-traumatic stress disorder: does gender matter? *European Journal of Psychotraumatology*, *9*(1), 1486123. https://doi.org/10.1080/20008198.2018.1486123
- Gallaway, M. S., Millikan, A. M., & Bell, M. R. (2011). The association between deployment-related posttraumatic growth among U.S. Army soldiers and negative behavioral health conditions. *J Clin Psychol*, 67(12), 1151-1160. https://doi.org/10.1002/jclp.20837
- Ganz, A., Yamaguchi, C., Parekh, B., Koritzky, G., & Berger, S. (2021). Military Culture and Its Impact on Mental Health and Stigma. *Journal of Community Engagement and Scholarship*, *13*(4). https://doi.org/10.54656/zzhp1245
- George, N., & Shepherd, L. J. (2016). Women, Peace and Security: Exploring the implementation and integration of UNSCR 1325. *International Political Science Review*, *37*(3), 297-306. https://doi.org/10.1177/0192512116636659
- Gewirtz, A. H., McMorris, B. J., Hanson, S., & Davis, L. (2014). Family adjustment of deployed and non-deployed mothers in families with a parent deployed to Iraq or Afghanistan. *Prof Psychol Res Pr*, 45(6), 465-477. https://doi.org/10.1037/a0036235
- Ghimire, B. (2017). Contributions and Challenges: Female Participations in UN Peacekeeping Missions. *International Journal of Humanities and Social Science*, 7, 207.
- Ghittoni, M., Lehouck, L., & Watson, C. (2018). Elsie Initiative for Women in Peace Operations: Baseline Study.
- Gibbons, S. W., Shafer, M., Aramanda, L., Hickling, E. J., & Benedek, D. M. (2014).

 Combat health care providers and resiliency: adaptive coping mechanisms during and after deployment. *Psychological services*, 11(2), 192-199.

 https://doi.org/10.1037/a0033165
- Gibbons, S. W., Shafer, M., Aramanda, L., Hickling, E. J., & Benedek, D. M. (2014).

 Combat health care providers and resiliency: Adaptive coping mechanisms during and after deployment. *Psychological services*, 11(2), 192.
- Gill, M. (2020). Phenomenology as qualitative methodology. In (pp. 73-94).



- Gist, R., & Woodall, S. J. (2013). There are no simple solutions to complex problems: The rise and fall of critical incident stress debriefing as a response to occupational stress in the fire service. In *Response to Disaster* (pp. 211-235). Routledge.
- Goldstein, L. A., Dinh, J., Donalson, R., Hebenstreit, C. L., & Maguen, S. (2017). Impact of military trauma exposures on posttraumatic stress and depression in female veterans. *Psychiatry Res*, 249, 281-285. https://doi.org/10.1016/j.psychres.2017.01.009
- Goodwin, L., & Rona, R. J. (2013). PTSD in the armed forces: what have we learned from the recent cohort studies of Iraq/Afghanistan? *J Ment Health*, 22(5), 397-401. https://doi.org/10.3109/09638237.2013.819422
- Gould, M., Adler, A., Zamorski, M., Castro, C., Hanily, N., Steele, N., Kearney, S., & Greenberg, N. (2010). Do stigma and other perceived barriers to mental health care differ across Armed Forces? *J R Soc Med*, 103(4), 148-156. https://doi.org/10.1258/jrsm.2010.090426
- Greenberg, N., Langston, V., & Gould, M. (2007). Culture: what is its effect on stress in the military? *Military medicine*, 172(9), 931. https://doi.org/https://doi.org/https://doi.org/10.7205/milmed.172.9.931
- Greenburg, J. (2017). New military femininities: humanitarian violence and the gendered work of war among U.S. servicewomen. *Gender, Place & Culture*, 24(8), 1107-1126. https://doi.org/10.1080/0966369X.2017.1347560
- Greene-Shortridge, T. M., Britt, T. W., & Castro, C. A. (2007). The stigma of mental health problems in the military. *Mil Med*, *172*(2), 157-161. https://doi.org/10.7205/milmed.172.2.157
- Greer, T. (2017). Career Development for Women Veterans: Facilitating Successful
 Transitions From Military Service to Civilian Employment. *Advances in Developing Human Resources*, 19, 54-65. https://doi.org/10.1177/1523422316682737
- Gruber, J., Prinstein, M. J., Clark, L. A., Rottenberg, J., Abramowitz, J. S., Albano, A. M.,
 Aldao, A., Borelli, J. L., Chung, T., Davila, J., Forbes, E. E., Gee, D. G., Hall, G. C.
 N., Hallion, L. S., Hinshaw, S. P., Hofmann, S. G., Hollon, S. D., Joormann, J.,
 Kazdin, A. E., . . . Weinstock, L. M. (2021). Mental health and clinical psychological
 science in the time of COVID-19: Challenges, opportunities, and a call to action.
 American Psychologist, 76(3), 409-426. https://doi.org/10.1037/amp0000707
- Halterman, A. L., & Irvine, J. A. (2014). Bounded altruism: INGOs' opportunities and constraints during humanitarian crises and the US intervention in Bosnia-Herzegovina



- and Kosovo. *East European Politics*, *30*(4), 458-481. https://doi.org/10.1080/21599165.2014.922468
- Han, S. C., Castro, F., Lee, L. O., Charney, M. E., Marx, B. P., Brailey, K., Proctor, S. P., & Vasterling, J. J. (2014). Military unit support, postdeployment social support, and PTSD symptoms among active duty and National Guard soldiers deployed to Iraq. *Journal of Anxiety Disorders*, 28(5), 446-453. https://doi.org/10.1016/j.janxdis.2014.04.004
- Hancock, D. R., Algozzine, B., & Lim, J. H. (2021). *Doing Case Study Research: A Practical Guide for Beginning Researchers*. Teachers College Press.

 https://books.google.co.za/books?id=e7lLEAAAQBAJ
- Harmse, A., O'Neil, S., & Strasheim, A. (2022). Organisational support to overcome the challenges of extended absences of officers in the SANDF. *Scientia Militaria*, 50(2). https://doi.org/10.5787/50-2-1374
- Harney, J., & Lerman, A. E. (2021). Clarifying the role of officer coping on turnover in corrections. *Criminal Justice Studies*, *34*(4), 397-422. https://doi.org/10.1080/1478601X.2021.1999117
- Hawker, D. M., Durkin, J., & Hawker, D. S. J. (2011). To debrief or not to debrief our heroes: that is the question. *Clinical Psychology & Psychotherapy*, *18*(6), 453-463. https://doi.org/https://doi.org/10.1002/cpp.730
- Heffer, T., & Willoughby, T. (2017). A count of coping strategies: A longitudinal study investigating an alternative method to understanding coping and adjustment. *PLoS One*, 12(10), e0186057. https://doi.org/10.1371/journal.pone.0186057
- Heinecken, L. (2015). Are Women 'Really' Making a Unique Contribution to Peacekeeping?: The Rhetoric and the Reality. *Journal of International Peacekeeping*, 19(3-4), 227-248. https://doi.org/https://doi.org/10.1163/18754112-01904002
- Heinecken, L. (2020). South Africa's post-apartheid military. Springer.
- Heinecken, L., & Ferreira, R. (2012). 'Fighting for peace'. *African Security Review*, 21(2), 50-60. https://doi.org/10.1080/10246029.2011.641688
- Heinecken, L., & Heinecken, L. (2020). Military Veterans: The Challenges of Reintegration and Compensation. South Africa's Post-Apartheid Military: Lost in Transition and Transformation, 87-103.
- Heinecken, L., & Wilén, N. (2021). No Place Like Home? Postdeployment Reintegration Challenges Facing South African Peacekeepers. *Armed Forces & Society*, 47(3), 415-434. https://doi.org/10.1177/0095327x19894719



- Helfrich, K. (2019). SANDF's Peace Mission Training Centre Enhancing African Peacekeeping', DefenceWeb, 4 July 2013. In.
- Hopkins, R. M., Regehr, G., & Pratt, D. D. (2017). A framework for negotiating positionality in phenomenological research(). *Med Teach*, *39*(1), 20-25. https://doi.org/10.1080/0142159x.2017.1245854
- Howard, L. M. (2019). *Power in peacekeeping*. Cambridge University Press.
- Hutton, P. L. (2022). *Humanitarian Fatigue: Towards a Definition* The Chicago School of Professional Psychology].
- Janoff-Bulman, R. (1989). Assumptive Worlds and the Stress of Traumatic Events: Applications of the Schema Construct. *Social Cognition*, 7(2), 113-136. https://doi.org/10.1521/soco.1989.7.2.113
- Johnson, J. L., Adkins, D., & Chauvin, S. (2020). A Review of the Quality Indicators of Rigor in Qualitative Research. *Am J Pharm Educ*, 84(1), 7120. https://doi.org/10.5688/ajpe7120
- Jones, C., Barrera, I., Brothers, S., Ring, R., & Wahlestedt, C. (2017). Oxytocin and social functioning. *Dialogues Clin Neurosci*, 19(2), 193-201. https://doi.org/10.31887/DCNS.2017.19.2/cjones
- Kabadayı Şahin, E., & Sevil, F. (2023). From Trauma to Post-Traumatic Stress Disorder: Identification of the Risk Factors. *Ankara Med J*, 23(2), 241-255. https://doi.org/10.5505/amj.2023.90699
- Karim, S. (2017). Reevaluating Peacekeeping Effectiveness: Does Gender Neutrality Inhibit Progress? *International Interactions*, 43(5), 822-847. https://doi.org/10.1080/03050629.2017.1231113
- Karim, S., & Beardsley, K. (2017). *Equal opportunity peacekeeping: women, peace, and security in post-conflict states*. Oxford University Press.
- Kearns, M., Ressler, K., Zatzick, D., & Rothbaum, B. (2012). Early interventions for PTSD: A review. *Depression and anxiety*, 29, 833-842. https://doi.org/10.1002/da.21997
- Klenke, K. (2017). Women's Leadership in Context. In *Women in Leadership 2nd Edition* (pp. 1-34). Emerald Publishing Limited.
- Kline, A., Ciccone, D. S., Weiner, M., Interian, A., Hill, L. S., Falca-Dodson, M., Black, C. M., & Losonczy, M. (2013). Gender Differences in the Risk and Protective Factors Associated With PTSD: A Prospective Study of National Guard Troops Deployed to Iraq. *Psychiatry: Interpersonal and Biological Processes*, 76(3), 256-272. https://doi.org/10.1521/psyc.2013.76.3.256



- Knapen, J., Vancampfort, D., Moriën, Y., & Marchal, Y. (2015). Exercise therapy improves both mental and physical health in patients with major depression. *Disabil Rehabil*, 37(16), 1490-1495. https://doi.org/10.3109/09638288.2014.972579
- Koopman, R., & Dyk, G. A. J. V. (2012). Peacekeeping operations and adjustment of soldiers in Sudan: Peace in the minds and hearts of soldiers? *African Journal on Conflict Resolution*, 12(3), 53-76. https://doi.org/doi:10.10520/EJC129468
- Korstjens, I., & Moser, A. (2018). Series: Practical guidance to qualitative research. Part 4: Trustworthiness and publishing. *European Journal of General Practice*, *24*(1), 120-124. https://doi.org/10.1080/13814788.2017.1375092
- Kreutz, J., & Cardenas, M. (2017). Women, peace and intervention: how the international community responds to sexual violence in civil conflict. *Canadian Foreign Policy Journal*, 23(3), 260-276. https://doi.org/10.1080/11926422.2017.1343197
- Langston, V., Gould, M., & Greenberg, N. (2007). Culture: What Is Its Effect on Stress in the Military? *Military medicine*, 172, 931-935.

 https://doi.org/10.7205/MILMED.172.9.931
- Lazarus, R. S. (1989). Psychological stress in the workplace. J uoeh, 11 Suppl, 528-540.
- Lazarus, R. S. (1993). Coping theory and research: past, present, and future. *Psychosom Med*, 55(3), 234-247. https://doi.org/10.1097/00006842-199305000-00002
- Lazarus, R. S. (2000). Evolution of a model of stress, coping, and discrete emotions.

 Handbook of stress, coping, and health: Implications for nursing research, theory, and practice, 195-222.
- Lazarus, R. S. (2020). Psychological stress in the workplace. In *Occupational stress* (pp. 3-14). CRC Press.
- Lazarus, R. S., & Folkman, S. (1984). *Stress, appraisal, and coping*. Springer publishing company.
- Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic Inquiry*. SAGE Publications. https://books.google.co.za/books?id=2oA9aWlNeooC
- Loscalzo, Y., Giannini, M., Gori, A., & Fabio, A. D. (2018). The Wellbeing of Italian Peacekeeper Military: Psychological Resources, Quality of Life and Internalizing Symptoms. *Frontiers in Psychology*, 9. https://doi.org/10.3389/fpsyg.2018.00103
- Love, B., Vetere, A., & Davis, P. (2020). Should Interpretative Phenomenological Analysis (IPA) be Used With Focus Groups? Navigating the Bumpy Road of "Iterative Loops," Idiographic Journeys, and "Phenomenological Bridges". *International Journal of Qualitative Methods*, 19. https://doi.org/10.1177/1609406920921600



- Luedke, A., Lewis, C., & Rodriguez, M. (2017). Sexual Violence, Exploitation, and Abuse. JSTOR.
- Mahamid, F. A., & Bdier, D. (2021). The Association Between Positive Religious Coping,
 Perceived Stress, and Depressive Symptoms During the Spread of Coronavirus
 (COVID-19) Among a Sample of Adults in Palestine: Across Sectional Study. *J Relig Health*, 60(1), 34-49. https://doi.org/10.1007/s10943-020-01121-5
- Mahindru, A., Patil, P., & Agrawal, V. (2023). Role of Physical Activity on Mental Health and Well-Being: A Review. *Cureus*, *15*(1), e33475. https://doi.org/10.7759/cureus.33475
- Marini, C. M., Wadsworth, S. M., Christ, S. L., & Franks, M. M. (2017). Emotion expression, avoidance and psychological health during reintegration: A dyadic analysis of actor and partner associations within a sample of military couples. *Journal of Social and Personal Relationships*, 34(1), 69-90. https://doi.org/10.1177/0265407515621180
- Markowitz, J. C., Petkova, E., Neria, Y., Van Meter, P. E., Zhao, Y., Hembree, E., Lovell, K., Biyanova, T., & Marshall, R. D. (2015). Is Exposure Necessary? A Randomized Clinical Trial of Interpersonal Psychotherapy for PTSD. *American Journal of Psychiatry*, 172(5), 430-440. https://doi.org/10.1176/appi.ajp.2014.14070908
- Martín de la Rosa, V., & Lázaro, L. M. (2019). How women are imagined through conceptual metaphors in United Nations Security Council Resolutions on women, peace and security. *Journal of Gender Studies*, 28(4), 373-386. https://doi.org/10.1080/09589236.2017.1331844
- Martinez, A. (2022). Mental Health Stigma in the Military Context. *Silicon Valley Sociological Review*, 20(1), 10.
- Martos Martínez, Á., Molero Jurado, M. D. M., Pérez-Fuentes, M., Barragán Martín, A. B., Márquez, M., & Gázquez Linares, J. (2021). Role of Personality and Positive and Negative Affects in Coping Strategies of Nurses: A Cross-Sectional Study. Frontiers in Psychology, 12. https://doi.org/10.3389/fpsyg.2021.682153
- Mashatola, N., & Bester, P. (2020). Posttraumatic growth: new opportunities for the South African Military. *Contemporary Issues*, 205.
- Mashishi, K. (2012). Cultural intelligence: SA army in peace operations. *Thaba Tshwane*, 15.
- Mattocks, K. M., Haskell, S. G., Krebs, E. E., Justice, A. C., Yano, E. M., & Brandt, C. (2012). Women at war: understanding how women veterans cope with combat and military sexual trauma. *Soc Sci Med*, *74*(4), 537-545. https://doi.org/10.1016/j.socscimed.2011.10.039



- Mbhele, P. E. (2021). Combat exposure as moderator between risk, protective factors and suicide ideation for South African Army Infantry soldiers University of the Free State].
- McCullough, C. (2016). Creating Canada's peacekeeping past. ubc Press.
- McKim, C. (2023). Meaningful Member-Checking: A Structured Approach to Member-Checking [Article]. *American Journal of Qualitative Research (AJQR)*, 7, 41+. https://link.gale.com/apps/doc/A750229643/AONE?u=anon~ef92fa06&sid=googleScholar&xid=7236a5ff
- McNally, R. J., & Frueh, B. C. (2013). Why are Iraq and Afghanistan War veterans seeking PTSD disability compensation at unprecedented rates? *J Anxiety Disord*, *27*(5), 520-526. https://doi.org/10.1016/j.janxdis.2013.07.002
- Merriam, S. B., & Grenier, R. S. (2019). *Qualitative Research in Practice: Examples for Discussion and Analysis*. Wiley.

 https://books.google.co.za/books?id=PL59DwAAQBAJ
- Michalopoulou, L. E., Welsh, J. A., Perkins, D. F., & Ormsby, L. (2017a). Stigma and Mental Health Service Utilization in Military Personnel: A Review of the Literature. *Military Behavioral Health*, *5*(1), 12-25. https://doi.org/10.1080/21635781.2016.1200504
- Michalopoulou, L. E., Welsh, J. A., Perkins, D. F., & Ormsby, L. (2017b). Stigma and Mental Health Service Utilization in Military Personnel: A Review of the Literature. *Military Behavioral Health*, 5, 12 25.
- Miller, R. M., Chan, C. D., & Farmer, L. B. (2018). Interpretative phenomenological analysis: A contemporary qualitative approach. *Counselor Education and Supervision*, *57*(4), 240-254. https://doi.org/10.1002/ceas.12114
- Mitchell, M. M., Gallaway, M. S., Millikan, A. M., & Bell, M. R. (2013). Combat exposure, unit cohesion, and demographic characteristics of soldiers reporting posttraumatic growth. *Journal of Loss and Trauma*, *18*(5), 383-395. https://doi.org/10.1080/15325024.2013.768847
- Mohajan, H. (2018). Qualitative research methodology in social sciences and related subjects.

 **Journal of Economic Development, Environment and People, 7, 23-48.

 https://doi.org/10.26458/jedep.v7i1.571
- Moore, T. M., Risbrough, V. B., Baker, D. G., Larson, G. E., Glenn, D. E., Nievergelt, C. M., Maihofer, A., Port, A. M., Jackson, C. T., Ruparel, K., & Gur, R. C. (2017). Effects of military service and deployment on clinical symptomatology: The role of trauma



- exposure and social support. *Journal of Psychiatric Research*, *95*, 121-128. https://doi.org/https://doi.org/10.1016/j.jpsychires.2017.08.013
- Morgan, J. K., Hourani, L., & Tueller, S. (2017). Health-Related Coping Behaviors and Mental Health in Military Personnel. *Mil Med*, *182*(3), e1620-e1627. https://doi.org/10.7205/milmed-d-16-00165
- Morgan, J. K., Hourani, L., & Tueller, S. (2017). Health-Related Coping Behaviors and Mental Health in Military Personnel. *Military medicine*, *182*(3-4), e1620-e1627. https://doi.org/10.7205/milmed-d-16-00165
- Motta, V., & Larkin, M. (2023). Absence of other and disruption of self: an interpretative phenomenological analysis of the meaning of loneliness in the context of life in a religious community. *Phenomenology and the Cognitive Sciences*, 22(1), 55-80. https://doi.org/10.1007/s11097-022-09832-8
- Mtshayisa, V. E., & Letšosa, R. (2019). Practical guidelines to ameliorate the effects of internal and external deployments on the marriages of soldiers. *HTS*: *Theological Studies*, 75(4), 1-7. https://doi.org/doi:10.4102/hts.v75i4.5641
- Murray, D., Milton, D., Green, J., & Bervoets, J. (2023). The Human Spectrum: A

 Phenomenological Enquiry within Neurodiversity. *Psychopathology*, *56*(3), 220-230.

 https://doi.org/10.1159/000526213
- Nations., U. (2020). *Women in peacekeeping*. Retrieved 30 June from https://peacekeeping.un.org/en/women-peacekeeping
- Neubauer, B. E., Witkop, C. T., & Varpio, L. (2019). How phenomenology can help us learn from the experiences of others. *Perspect Med Educ*, 8(2), 90-97. https://doi.org/10.1007/s40037-019-0509-2
- Newby, V. F., & O'Malley, A. (2021). Introduction: WPS 20 Years On: Where Are the Women Now? *Global Studies Quarterly*, *1*(3), ksab017.
- Ngozwana, N. (2018). Ethical Dilemmas in Qualitative Research Methodology: Researcher's Reflections. *International Journal of Educational Methodology*, *4*(1), 19-28. https://doi.org/10.12973/ijem.4.1.19
- Nkewu, Z. (2014). Impact of psychological wellbeing and perceived combat readiness on willingness to deploy in the SANDF: an exploratory study.
- Nkewu, Z., & Van Dyk, G. (2016). Combat readiness: Perceived influences on willingness to deploy. *Journal of Psychology in Africa*, 26, 81-83. https://doi.org/10.1080/14330237.2015.1124618



- Nowell, L. S., Norris, J. M., White, D. E., & Moules, N. J. (2017). Thematic Analysis:Striving to Meet the Trustworthiness Criteria. *International Journal of Qualitative Methods*, *16*(1), 1609406917733847. https://doi.org/10.1177/1609406917733847
- Ntatamala, I., & Adams, S. (2022). The Correlates of Post-Traumatic Stress Disorder in Ambulance Personnel and Barriers Faced in Accessing Care for Work-Related Stress. *Int J Environ Res Public Health*, 19(4). https://doi.org/10.3390/ijerph19042046
- Nygaard, E., & Heir, T. (2012). World assumptions, posttraumatic stress and quality of life after a natural disaster: A longitudinal study. *Health and Quality of Life Outcomes*, 10(1), 76. https://doi.org/10.1186/1477-7525-10-76
- Nzabamwita Musagara, E. (2017). Battallian stress management strategic plan in operational areas role of clinical psychologist. Case of rwanbatt 37 in Sudan University of Rwanda].
- O'Reilly, M., Súilleabháin, A. Ó., & Paffenholz, T. (2022). Reimagining peacemaking: Women's roles in peace processes.
- Ochu, A. C., Davis, E. B., Magyar-Russell, G., O'Grady, K. A., & Aten, J. D. (2018).

 Religious coping, dispositional forgiveness, and posttraumatic outcomes in adult survivors of the Liberian Civil War. *Spirituality in Clinical Practice*, *5*(2), 104-119.

 https://doi.org/10.1037/scp0000163
- Ogińska-Bulik, N. (2017). Negative and positive effects of exposition to trauma among soldiers participating in military missions The role of rumination. *Postępy Psychiatrii i Neurologii*, 26, 1-12. https://doi.org/10.5114/ppn.2017.68290
- Olmos-Vega, F. M., Stalmeijer, R. E., Varpio, L., & Kahlke, R. (2023). A practical guide to reflexivity in qualitative research: AMEE Guide No. 149. *Medical Teacher*, 45(3), 241-251. https://doi.org/10.1080/0142159X.2022.2057287
- Owuor, V. (2021). The Role of United Nations Peacekeeping Female Engagement Teams in Peace Support Operations in Africa: a Case Study of Democratic Republic of Congo University of Nairobi].
- Panjwani, U., Dudani, S., & Wadhwa, M. (2021). Sleep, Cognition, and Yoga. *Int J Yoga*, 14(2), 100-108. https://doi.org/10.4103/ijoy.IJOY_110_20
- Park, C., Smith, P., Lee, S., Mazure, C., McKee, S., & Hoff, R. (2016). Positive and Negative Religious/Spiritual Coping and Combat Exposure as Predictors of Posttraumatic Stress and Perceived Growth in Iraq and Afghanistan Veterans. *Psychology of Religion and Spirituality*, 9. https://doi.org/10.1037/rel0000086



- Patnaik, E. (2013). Reflexivity: Situating the researcher in qualitative research. *Humanities*and Social Science Studies, 2, 98-106.

 https://doi.org/https://www.researchgate.net/publication/263916084_Reflexivity_Situating the researcher in qualitative research
- Peltzer, K., & Pengpid, S. (2019). High physical activity is associated with post-traumatic stress disorder among individuals aged 15 years and older in South Africa. *S Afr J Psychiatr*, 25, 1329. https://doi.org/10.4102/sajpsychiatry.v25i0.1329
- Pietkiewicz, I., & Smith, J. (2014). A practical guide to using Interpretative

 Phenomenological Analysis in qualitative research psychology. *CPPJ*, 20, 7-14.

 https://doi.org/10.14691/CPPJ.20.1.7
- Platania, S., Gruttadauria, S. V., & Morando, M. (2022). Dispositional Resilience as Mediator in Psychological Stress on Healthcare Workers: A Multi-Group Analysis of Frontline and Non-Frontline Workers. *Eur J Investig Health Psychol Educ*, *12*(9), 1285-1299. https://doi.org/10.3390/ejihpe12090089
- Polusny, M. A., Erbes, C. R., Thuras, P., Moran, A., Lamberty, G. J., Collins, R. C., Rodman, J. L., & Lim, K. O. (2015). Mindfulness-Based Stress Reduction for Posttraumatic Stress Disorder Among Veterans: A Randomized Clinical Trial. *JAMA*, 314(5), 456-465. https://doi.org/10.1001/jama.2015.8361
- Possemato, K., McKenzie, S., McDevitt-Murphy, M. E., Williams, J. L., & Ouimette, P. (2014). The Relationship Between Postdeployment Factors and PTSD Severity in Recent Combat Veterans. *Military Psychology*, 26(1), 15-22. https://doi.org/10.1037/mil0000027
- Qu, S., & Dumay, J. (2011). The qualitative research interview. *Qualitative Research in Accounting & Management*, 8, 238-264. https://doi.org/10.1108/11766091111162070
- Rahman, M. (2016). The Advantages and Disadvantages of Using Qualitative and Quantitative Approaches and Methods in Language "Testing and Assessment" Research: A Literature Review. *Journal of Education and Learning*, 6, 102. https://doi.org/10.5539/jel.v6n1p102
- Rajappa, K., Gallagher, M., & Miranda, R. (2012). Emotion Dysregulation and Vulnerability to Suicidal Ideation and Attempts. *Cognitive Therapy and Research*, *36*(6), 833-839. https://doi.org/10.1007/s10608-011-9419-2
- Ramsook, L. (2018). A Methodological Approach to Hermeneutic Phenomenology.



- Reid, A.-M., Brown, J., Smith, J., Cope, A., & Jamieson, S. (2018). Ethical dilemmas and reflexivity in qualitative research. *Perspectives on Medical Education*, 7. https://doi.org/10.1007/s40037-018-0412-2
- Resick, P. A., Monson, C. M., & Chard, K. M. (2017). *Cognitive processing therapy for PTSD: A comprehensive manual*. The Guilford Press.
- Rhoads, E. P. (2016). *Taking sides in peacekeeping: impartiality and the future of the United Nations*. Oxford University Press.
- Richardson, J. D., King, L., Sareen, J., & Elhai, J. D. (2016). Post-traumatic stress symptoms 5 years after military deployment to Afghanistan. *Lancet Psychiatry*, *3*(1), 7-9. https://doi.org/10.1016/s2215-0366(15)00418-6
- Rickovic, V. (2014). Stress Factors and Positive Aspects of Deployment to International Military Operation. *Journal of Psychology & Clinical Psychiatry*, *I*(7). https://doi.org/10.15406/jpcpy.2014.01.00049
- Romero, D. H., Riggs, S. A., Raiche, E., McGuffin, J., & Captari, L. E. (2020). Attachment, coping, and psychological symptoms among military veterans and active duty personnel. *Anxiety, Stress, & Coping*, *33*(3), 326-341. https://doi.org/10.1080/10615806.2020.1723008
- Rose, S. C., Bisson, J., Churchill, R., & Wessely, S. (2002). Psychological debriefing for preventing post traumatic stress disorder (PTSD). *Cochrane Database of Systematic Reviews*(2). https://doi.org/10.1002/14651858.CD000560
- Roulston, K. (2010). *Reflective Interviewing: A Guide to Theory and Practice*. SAGE Publications. https://books.google.co.za/books?id=YIfHXeTDsTIC
- Rowland Dom, A., Anthonia Chigozie, N., & Obiageli Felicia, O. (2021). Social Support and Resilience Predicting Post-Traumatic Stress Disorder among Returned Soldiers from Insurgency Areas in Nigeria. *ESUT JOURNAL OF SOCIAL SCIENCES*, *6*(3). https://esutjss.com/index.php/ESUTJSS/article/view/80
- Sareen, J. (2014). Posttraumatic Stress Disorder in Adults: Impact, Comorbidity, Risk Factors, and Treatment. *The Canadian Journal of Psychiatry*, *59*(9), 460-467. https://doi.org/10.1177/070674371405900902
- Saunders, B., Kitzinger, J., & Kitzinger, C. (2015). Anonymising interview data: challenges and compromise in practice. *Qual Res*, 15(5), 616-632. https://doi.org/10.1177/1468794114550439
- Schumm, J., & Chard, K. (2012). Alcohol and Stress in the Military. *Alcohol research*: current reviews, 34, 401-407.



- Seelandt, J. C., Walker, K., & Kolbe, M. (2021). "A debriefer must be neutral" and other debriefing myths: a systemic inquiry-based qualitative study of taken-for-granted beliefs about clinical post-event debriefing. *Advances in Simulation*, 6(1), 7. https://doi.org/10.1186/s41077-021-00161-5
- Sharing the front line and the back hills: International protectors and providers:

 Peacekeepers, humanitarian aid workers and the media in the midst of crisis. (2002).

 Baywood Publishing Co.
- Sharland, L. (2021). Women, peace, and security mandates for UN peacekeeping operations: assessing influence and impact.
- Shelef, L., Fruchter, E., Gordon, S., Garbi, D., Nirit, Y., Haber, Y., Ginat, K., Oz, I., & Shahar, G. (2018). Impairment among soldiers who were exposed to combat during military operation. *International Journal of Psychiatry*, *3*(1), 1-5.
- Shi, L., Wang, L., Jia, X., Li, Z., Mu, H., Liu, X., Peng, B., Li, A., & Fan, L. (2017).

 Prevalence and correlates of symptoms of post-traumatic stress disorder among

 Chinese healthcare workers exposed to physical violence: a cross-sectional study.

 BMJ Open, 7(7), e016810. https://doi.org/10.1136/bmjopen-2017-016810
- Shinga, G. N. (2015). Factors involved in combat readiness with hardiness as a mediator: an exploratory study Stellenbosch: Stellenbosch University].
- Sibanda, S. (2020). An object relations perspective on accounts of traumatisation among a group of Black South African National Defence Force soldiers University of Pretoria (South Africa)].
- Simon, N., Roberts, N. P., Lewis, C. E., van Gelderen, M. J., & Bisson, J. I. (2019).

 Associations between perceived social support, posttraumatic stress disorder (PTSD) and complex PTSD (CPTSD): implications for treatment. *European Journal of Psychotraumatology*, 10(1), 1573129.

 https://doi.org/10.1080/20008198.2019.1573129
- Singh Thakuri, B. (2020). Prevalence and Factors Affecting the Stress among Nepalese Peacekeepers
- Smith, J. A. (2011). Evaluating the contribution of interpretative phenomenological analysis. *Health Psychology Review*, 5(1), 9-27. https://doi.org/10.1080/17437199.2010.510659
- Smith, J. A., & Osborn, M. (2007). Pain as an assault on the self: An interpretative phenomenological analysis of the psychological impact of chronic benign low back



- pain. *Psychology & Health*, 22(5), 517-534. https://doi.org/10.1080/14768320600941756
- Smith, J. A., & Osborn, M. (2015). Interpretative phenomenological analysis as a useful methodology for research on the lived experience of pain. *British Journal of Pain*, 9(1), 41-42. https://doi.org/10.1177/2049463714541642
- Smith, M. M., Saklofske, D. H., Keefer, K. V., & Tremblay, P. F. (2016). Coping Strategies and Psychological Outcomes: The Moderating Effects of Personal Resiliency. *The Journal of Psychology*, *150*(3), 318-332. https://doi.org/10.1080/00223980.2015.1036828
- Smith, P. B., Easterbrook, M. J., Celikkol, G. C., Chen, S. X., Ping, H., & Rizwan, M. (2016). Cultural Variations in the Relationship Between Anger Coping Styles, Depression, and Life Satisfaction. *Journal of Cross-Cultural Psychology*, 47(3), 441-456. https://doi.org/10.1177/0022022115620488
- Smith, T. C., Ryan, M. A. K., Wingard, D. L., Slymen, D. J., Sallis, J. F., & Kritz-Silverstein, D. (2008). New onset and persistent symptoms of post-traumatic stress disorder self reported after deployment and combat exposures: prospective population based US military cohort study. *BMJ*, *336*(7640), 366-371. https://doi.org/10.1136/bmj.39430.638241.AE
- Snowdon, K. (2021). Exploring the clinical debrief: benefits and barriers. *Journal of Paramedic Practice*, 13(1), 1-7. https://doi.org/10.12968/jpar.2021.13.CPD1
- Soir, E. (2017). Psychological Adjustment After Military Operations: The Utility of Postdeployment Decompression for Supporting Health Readjustment. In (pp. 89-103). https://doi.org/10.1007/978-3-319-66192-6_7
- Solomon, Z. (2020). From the Frontline to the Homefront: The Experience of Israeli Veterans [Review]. *Frontiers in Psychiatry*, 11. https://doi.org/10.3389/fpsyt.2020.589391
- Souza, W. F., Figueira, I., Mendlowicz, M. V., Volchan, E., Portella, C. M., Mendonça-de-Souza, A. C. F., & Coutinho, E. S. F. (2011). Posttraumatic Stress Disorder in Peacekeepers: A Meta-Analysis. *The Journal of Nervous and Mental Disease*, 199(5). https://journals.lww.com/jonmd/fulltext/2011/05000/posttraumatic_stress_disorder_in_peacekeepers_a.4.aspx
- Stahl, N., & King, J. (2020). Expanding Approaches for Research: Understanding and Using Trustworthiness in Qualitative Research. *44*, 26-28.



- Stanisławski, K. (2019). The Coping Circumplex Model: An Integrative Model of the Structure of Coping With Stress. *Frontiers in Psychology*, 10. https://doi.org/10.3389/fpsyg.2019.00694
- Steidl, C. R., & Brookshire, A. R. (2019). Just one of the guys until shower time: How symbolic embodiment threatens women's inclusion in the US military. *Gender, Work & Organization*, 26(9), 1271-1288.

 https://doi.org/https://doi.org/10.1111/gwao.12320
- Stenfors, T., Kajamaa, A., & Bennett, D. (2020). How to ... assess the quality of qualitative research. *Clin Teach*, 17(6), 596-599. https://doi.org/10.1111/tct.13242
- Stępka, M. (2018). Stepka Maciej. 2018. Humanitarian Securitization of the 2015 'Migration Crisis'. Investigating Humanitarianism and Security in the EU Policy Frames on Operational Involvement in the Mediterranean," in Migration Policy in Crisis, ed. Ibrahim Sirkeci et al. London, 9–30. In (pp. 9-30).
- Tamrakar, T., Murphy, J., & Elklit, A. (2019a). Was psychological debriefing dismissed too quickly? *Crisis, Stress, and Human Resilience: An International Journal*, 1(3), 146-155.
- Tamrakar, T., Murphy, J., & Elklit, A. (2019b). Was Psychological Debriefing Dismissed Too Quickly?: Assessment of the 2002 Cochrane Review.
- Tan, L., Petrie, K., Deady, M., Bryant, R. A., & Harvey, S. B. (2022). Systematic review of first responder post-deployment or post-incident psychosocial interventions. *Occupational Medicine*, 72(3), 160-169. https://doi.org/10.1093/occmed/kqab182
- Taylor-Desir, M., & Association, A. P. (2022). What is Posttraumatic Stress Disorder (PTSD). American Psychiatric Association. https://www.psychiatry.org/patients-families/ptsd/what-is-ptsd
- Tedeschi, R. G., & Calhoun, L. G. (1996). The Posttraumatic Growth Inventory: Measuring the positive legacy of trauma. *Journal of Traumatic Stress*, 9(3), 455-472. https://doi.org/10.1002/jts.2490090305
- Tedeschi, R. G., Shakespeare-Finch, J., Taku, K., & Calhoun, L. G. (2018). Posttraumatic Growth. https://doi.org/10.4324/9781315527451
- Thomassen, Å. G., Hystad, S. W., Johnsen, B. H., Johnsen, G. E., & Bartone, P. T. (2018).

 The effect of hardiness on PTSD symptoms: A prospective mediational approach. *Military Psychology*, 30(2), 142-151. https://doi.org/10.1080/08995605.2018.1425065



- Tracy, S. J., & Hinrichs, M. M. (2017). Big Tent Criteria for Qualitative Quality. *The International Encyclopedia of Communication Research Methods*, 1-10. https://doi.org/10.1002/9781118901731.iecrm0016
- Trobaugh, E. M. (2018). Women, regardless: understanding gender bias in US military integration. *Joint Force Quarterly*, 88(1), 46-53.
- Tuckey, M. R., & Scott, J. E. (2014). Group critical incident stress debriefing with emergency services personnel: a randomized controlled trial. *Anxiety, Stress, & Coping*, 27(1), 38-54. https://doi.org/10.1080/10615806.2013.809421
- Tufford, L., & Newman, P. (2010). Bracketing in Qualitative Research. *Qualitative Social Work*, 11, 80-96. https://doi.org/10.1177/1473325010368316
- Tuffour, I. (2017). A Critical Overview of Interpretative Phenomenological Analysis: A Contemporary Qualitative Research Approach. *Journal of Healthcare Communications*, 02. https://doi.org/10.4172/2472-1654.100093
- Using Visual Data in Qualitative Research. (2018). SAGE Publications Ltd. http://digital.casalini.it/9781526426116
- http://digital.casalini.it/5018943
- Vahedi, L., Bartels, S. A., & Lee, S. (2021). 'Even peacekeepers expect something in return': A qualitative analysis of sexual interactions between UN peacekeepers and female Haitians. *Global Public Health*, *16*(5), 692-705. https://doi.org/10.1080/17441692.2019.1706758
- van Heerden, A. (2017). Military Psychology for Africa. GAJ van Dyk. *Scientia Militaria*, 44. https://doi.org/10.5787/44-2-1182
- van't Wout, C., & van Dyk, G. (2016). a psyChologiCal plan for miliTary operaTions in afriCa. *Military Psychology for Africa*, 67.
- Varkey, B. (2021). Principles of Clinical Ethics and Their Application to Practice. *Medical Principles and Practice*, 30(1), 17-28. https://doi.org/10.1159/000509119
- Vermeij, L. (2022). Woman first, soldier second: Taboos and stigmas facing military women in UN peace operations. International Peace Institute.
- Vermeij, L., & Giannini, R. (2014). Women, peace and security: gender challenges within UN peacekeeping missions. https://doi.org/http://hdl.handle.net/11250/278040
- Vignaud, P., Lavallé, L., Brunelin, J., & Prieto, N. (2022). Are psychological debriefing groups after a potential traumatic event suitable to prevent the symptoms of PTSD? *Psychiatry Research*, *311*, 114503.
 - https://doi.org/https://doi.org/10.1016/j.psychres.2022.114503



- Visagie, N., Toit, R. A. D., Joubert, S., Schoeman, D., & Zungu, D. (2022). Comparing deployment experiences of South African National Defence Force personnel during peace support missions: Sudan vs Democratic Republic of Congo. *Scientia Militaria*: South African Journal of Military Studies, 50(2), 127-154.

 https://doi.org/doi:10.5787/50-2-1375
- Vogt, D., Vaughn, R., Glickman, M. E., Schultz, M., Drainoni, M. L., Elwy, R., & Eisen, S. (2011). Gender differences in combat-related stressors and their association with postdeployment mental health in a nationally representative sample of U.S. OEF/OIF veterans. *J Abnorm Psychol*, 120(4), 797-806. https://doi.org/10.1037/a0023452
- Vong, S. K., Kang, L., & Carter, S. R. (2022). Consumers' self-reported adherence to directions for non-prescription medicines and the role of risk perception. *Res Social Adm Pharm*, 18(11), 3929-3938. https://doi.org/10.1016/j.sapharm.2022.06.004
- Waller, M., Treloar, S. A., Sim, M. R., McFarlane, A. C., McGuire, A. C. L., Bleier, J., & Dobson, A. J. (2012). Traumatic events, other operational stressors and physical and mental health reported by Australian Defence Force personnel following peacekeeping and war-like deployments. *BMC Psychiatry*, 12. https://doi.org/10.1186/1471-244X-12-88
- Wang, Y., Chung, M. C., Wang, N., Yu, X., & Kenardy, J. (2021). Social support and posttraumatic stress disorder: A meta-analysis of longitudinal studies. *Clinical Psychology Review*, 85, 101998.
 https://doi.org/https://doi.org/10.1016/j.cpr.2021.101998
- Warner, C. (2018). Humanitarian and Peacekeeping Deployments. *American Military Life in the 21st Century: Social, Cultural, and Economic Issues and Trends*, *1*, 150-159.
- Wastell, C. (2020). *Understanding trauma and emotion: Dealing with trauma using an emotion-focused approach*. Routledge.
- Watson, C. (2019). Defence and gender. DCAF.
- Wegner, B. (2011). The difficult reintegration of soldiers to society and family after deployment. *ESSAI*, *9*(1), 41.
- Welsh, J., Olson, J., Perkins, D., Travis, W., & Ormsby, L. (2015). The Role of Natural Support Systems in the Post-deployment Adjustment of Active Duty Military Personnel. *American journal of community psychology*, *56*. https://doi.org/10.1007/s10464-015-9726-y
- Wessely, S., Rose, S., & Bisson, J. (2000). Brief psychological interventions ("debriefing") for trauma-related symptoms and the prevention of post traumatic stress disorder.



- *Cochrane Database Syst Rev*, 2002(2), Cd000560. https://doi.org/10.1002/14651858.Cd000560
- WHO. (2022). *Mental Disorders*. Retrieved 6 June from https://www.who.int/news-room/fact-sheets/detail/mental-disorders
- Wild, J., Greenberg, N., Moulds, M. L., Sharp, M.-L., Fear, N., Harvey, S., Wessely, S., & Bryant, R. A. (2020). Pre-incident training to build resilience in first responders: Recommendations on what to and what not to do. *Psychiatry: Interpersonal and Biological Processes*, 83(2), 128-142. https://doi.org/10.1080/00332747.2020.1750215
- Wilén, N. (2020). Female peacekeepers' added burden. *International Affairs*, 96(6), 1585-1602. https://doi.org/10.1093/ia/iiaa132
- Wilén, N., & Heinecken, L. (2017). Peacekeeping deployment abroad and the self-perceptions of the effect on career advancement, status and reintegration.

 International Peacekeeping, 24(2), 236-253.

 https://doi.org/10.1080/13533312.2016.1215247
- Willig, C. (2013). *Introducing Qualitative Research In Psychology*. Open University Press. https://books.google.co.za/books?id=E-lhuM-pNV8C
- Wisén, N., Larsson, G., Arborelius, U., & Risling, M. (2021). Are Peacekeeping Missions Inevitably Stressful? *Scandinavian Journal of Military Studies*. https://doi.org/10.31374/sjms.107
- Wood, M. D., Foran, H. M., & Britt, T. W. (2023). Limitations of benefit finding as a coping mechanism for combat-related PTSD symptoms. *Military Psychology*, *35*(3), 233-244. https://doi.org/10.1080/08995605.2022.2112884
- Wright, B. K., Kelsall, H. L., Sim, M. R., Clarke, D. M., & Creamer, M. C. (2013). Support Mechanisms and Vulnerabilities in Relation to PTSD in Veterans of the Gulf War, Iraq War, and Afghanistan Deployments: A Systematic Review. *Journal of Traumatic Stress*, 26(3), 310-318. https://doi.org/https://doi.org/10.1002/jts.21809
- Yavnai, N., Oz, I., Shahar, G., Gordon, S., Garbi, D., & Shelef, L. (2018). Impairment among Soldiers Who were Exposed to Combat During Military Operation. *International Journal of Psychiatry*, 3, 1-5.
- Zhao, X., Wang, J., & Shi, C. (2020). The Influence of Mental Resilience on the Positive Coping Style of Air Force Soldiers: A Moderation- Mediation Model. *Frontiers in Psychology*, 11. https://doi.org/10.3389/fpsyg.2020.00550



Zungu, D., & Visagie, N. (2020). All eyes on Sudan: The journey of female psychologists in the theatre of operation. SA Journal of Industrial Psychology, 46. https://doi.org/10.4102/sajip.v46i0.1739



Appendices

Appendix A- Written consent to participate in this study



Appendix B - Participant information sheet



Appendix C- Interview Schedule



Appendix D– Scripts of participants













Appendix E– Representation of master theme table



Representation of master themes and s

Appendix F- Extracts from scripts



Appendix G - Turnitin Report





$Appendix \ H-Ethics \ Approval$

